# Chemical Legacy Contamination of the Child

Catherine N. Dorey, PhD.

We are guilty of many errors and faults, but our worst crime is abandoning the children, neglecting the foundation of life. Many of the things we need can wait. The child cannot. Right now is the time his bones are being formed, his blood is being made and his senses are being developed. To him we cannot answer 'Tomorrow'. His name is 'Today'.

**Gabriela Mistral,** 1945 Nobel Prize Winner for Literature

This stuff is everywhere.

John Jake Ryan, Health Canada, Ottawa, 2001

## **Contents**

**List of Tables and Figures** 

Acknowledgements

**Preface** 

3.2.6 Metabolism

3.2.7 Biotransformation of Chemicals

3.2.8 Excretion and Elimination

Exe	cutive Summary	6			
1	INTRODUCTION	8	4	CONTAMINATING CHEMICALS AND DISEASE	24
2	CHEMICAL CONTAMINANTS				
	IN CHILDREN	10	4.1	Diseases on the Rise	24
			4.1.1	Infant Mortality	24
2.1	Alkylphenols	10	4.1.2	Immune Diseases	24
2.1.1	Alkylphenols in the Child	10	4.1.3	Cancers	25
2.2	Bisphenol A	10	4.1.4	Nervous System Disorders	26
2.2.1	Bisphenol A in the Child	11	4.1.5	Developmental and	
2.3	Brominated Flame Retardants	12		Reproductive Disorders	26
2.3.1	Brominated Flame Retardants in the Child	12			
2.4	Organotins	13	4.2	Vulnerable Stages	
2.4.1	Organotins in the Child	14		in Developing Cells	27
2.5	Phthalates	14	4.2.1	Control of Cell Division	27
2.5.1	Phthalates in the Child	14	4.2.2	Programmed Cell Death	27
2.6	Artificial Musks	16	4.2.3	Gene Expression	27
2.6.1	Artificial Musks in the Child	17	4.2.4	Cell Differentiation and Maturation	28
2.7	Chlorinated Paraffins	17			
2.7.1	Chlorinated Paraffins in the Child	17	4.3	Vulnerable Stages	
				in Child Development	28
			4.3.1	Germ Cell Development	28
3	CHILD EXPOSURE		4.3.2	Embryonic and Foetal Development	28
	TO CHEMICALS	18	4.3.3	Infancy and Childhood	29
			4.3.4	Puberty	29
3.1	From Products to Children	18			
3.1.1	Prenatal Exposure	18	4.4	Health Effects of	
3.1.2	Postnatal Exposure	19		Contaminating Chemicals	29
			4.4.1	Alkylphenols	30
3.2	Increased Exposure in Children	21	4.4.2	Bisphenol A	31
3.2.1	Diet	21	4.4.3	Brominated Flame Retardants	33
3.2.2	Behaviour	21	4.4.4	Organotins	34
3.2.3	Stature	22	4.4.5	Phthalates	35
3.2.4	Enhanced absorption	22	4.4.6	Artificial Musks	37
3.2.5	Distribution of in the body	22	4.4.7	Chlorinated Paraffins	38

23

23

5

6

**CONCLUSION** 

**REFERENCES** 

39

41

2



## List of Tables and Figures

Table 2.1.	Human contamination by nonylphenol.
Table 2.2.	Human contamination by bisphenol A.
Table 2.3.	Human contamination by polybrominated diphenyl ethers.
Table 2.4.	Human contamination by organotins.
Table 2.5.	Human contamination of blood from adults by the phthalate DEHP.
Table 2.6.	Contamination of blood from normal girls and girls with premature breast development (thelarche) by phthalates.
Table 2.7.	Human contamination by artifical musks.
Table 3.1.	A comparison of the estimated daily intake of the phthalate, DEHP, from varying sources by different age groups in Canada.
Table 3.2.	Examples of nonylphenol levels in infant foods, and general foods that may be fed to infants.
Table 3.3.	Energy intake of children at selected ages.
Table 4.1.	Infant mortality rates and percentage of total infant deaths for the ten leading causes of infant deaths in 1999 in the USA.
Table 4.2.	Reproductive effects of 4-octylphenol following prenatal and postnatal exposure via drinking water.
Table 4.3.	Reproductive effects of nonylphenol following multigenerational exposure via drinking water.
Table 4.4.	Assumed safe dose of bisphenol A compared to evidence of low-dose toxicity in rodents.
Table 4.5.	Different phthalate metabolites can both stimulate and suppress antibody production depending on their concentration levels.
Figure 4.1.	The structural similarity of the brominated flame retardants, polybrominated diphenyl ethers (PBDEs), to the polychlorinated bisphenols (PCBs) and thyroid hormones may underlie their toxicity.

## **Acknowledgements**

Many thanks to the following people for their assistance with this report:

Oliver Knowles and Mark Strutt (Toxic Campaigners, Greenpeace) for providing me with the opportunity to write about such an interesting, if not worrying, topic; Dr Vyvyan Howard, (Developmental Toxico-Pathologist, University of Liverpool) for his review and preface, and for his interesting discussion on evolving theories of cancer; Dr Doug Parr (Chief Scientist, Greenpeace) for his review and comments; and Chris Suttie (Researcher) for helping to track down rare articles on the contamination of children by persistent organic chemicals.

Finally, thanks to Sandra Steingraber whose book *Having Faith: An Ecologist's Journey to Motherhood*, gave me invaluable insights into the fascinating world of pregnancy and the developing child, and the potentially devastating effects of chemicals on child health.

### **Preface**

In writing Chemical Legacy: Contamination of the Child, Dr Dorey has performed a considerable service. Children's health is clearly an emotive topic, but this is a scholarly report of the current state of knowledge about the contamination of the human species by a restricted list of chemical pollutants. The data come from reputable published sources, primarily peer-reviewed journals and government publications. The interpretation of the findings is guarded and dispassionate. This publication will serve as a useful literature review.

To set the report in context, I welcome the opportunity to discuss some aspects of environmental pollution and child health.

#### **Evolution**

Throughout evolution, organisms have, by chance mutations, produced a succession of novel compounds that have improved their own survival prospects at the expense of other species. An excellent example is penicillin, produced by yeasts to inhibit bacteria. Thus the 'chemical warfare' concept in nature throughout evolution is not a new one. However, when such changes occur, they do so on a very local basis and in low volume. Furthermore, the target species either adapts or succumbs to the new threat. The adaptation is usually realised by a progressively increasing ability of the target species to metabolise and detoxify the novel compound. In higher creatures, such as mammals, this process might be expected to occur over hundreds of generations.

During the past few decades, the human race, through its industries, has managed to disperse high volumes of many novel chemical pollutants throughout the biosphere at concentrations high enough to cause adverse biological effects.

It should come as no surprise to discover

that some manufactured chemical compounds, many of which are surprisingly similar in structure to natural hormones, interfere with the chemical signalling systems within our own bodies. The control of development in the foetus and infant is very much under the control of hormones which are active in low levels of parts per trillion. This is within the range of environmental concentrations found for many environmental pollutants within the general population.

#### **Toxicology**

The mixture of chemicals that is found in the average person is extremely complex, in the toxicological sense. There are hundreds of groups of chemicals to be found within most people in the general population. Furthermore, within each group of organic chemicals there are usually many variants. For example, consider the pesticide toxaphene: there are over 60,000 congeners and enantiomers (mirror image molecules) possible from that single chemical group (Liem & Theelen, 1997; Vetter & Luckas, 1995; Vetter & Scherer, 1998). Then, in addition, there are metabolites to add to the complexity.

The realistic bottom line is that we are being exposed to tens of thousands of chemicals which simply didn't exist on this planet until a few decades ago. When our grandparents were in their mothers' wombs they would not have been exposed to these novel chemicals.

We have no tools for analysing the toxicity of the complex mixture to which we are exposed (Howard, 1997; Lang, 1995). Most toxicology is performed on chemicals one at a time. We have difficulty comparing chemicals even in combinations of pairs (Axelrad et al., 2002 and 2003). When we consider combinations of hundreds or thousands of chemicals, it is clear that we have not yet developed any methods that will allow for their rigorous toxicological analysis.



There is no indication of any imminent major breakthrough to this testing problem. That leaves us with only a generalised approach to safety through hazard reduction, i.e. reduction in exposure, using a precautionary approach. There are simply no other rational options available.

#### **Epidemiology**

To perform a satisfactory epidemiological study you need to know where you started from, i.e. the baseline condition. This allows for comparison with the exposed group. You also need to know about exposure – who is exposed to what – then you can measure exposure against outcome. As Dr Dorey's report makes clear, there is precious little exposure data for most of the groups of chemicals of concern. Furthermore, there are no unexposed groups to act as 'controls'. Under such circumstances, epidemiology can only be regarded as an extremely blunt tool.

The frequency of a condition in the population is also a factor to be considered. If thalidomide had not caused phocomelia (limb shortening, a relatively obvious and rare condition), but had instead caused cleft palate (a relatively common condition), the likelihood is that we still would not know of the side effects of this drug. If the chemical pollutant mixture to which we are now exposed is causing changes in the frequency of relatively common conditions such as allergy and cancer, then epidemiology is likely to be of little help, given the data gaps outlined above. Once again, we are faced with the single option of hazard reduction as a means of regulation.

#### Conclusions

This report strengthens the evidence base relating human exposure to environmental chemical pollutants, particularly to those members of the population most at risk, namely children. We have no toxicological tools with which to analyse the level of complexity of the mixture to which we are exposed. We have no realistic prospect of being able to use epidemiology to help analyse the effects of multiple components of the mixture. However, we do know that many of

the chemicals in this mixture are indeed toxic.

In reality, the only effective remedy available to us is to reduce the exposure of the general population, on a precautionary basis, by removing from production those groups of chemicals which have been shown to potentially pose a hazard.

C Vyvyan Howard MB. ChB. PhD. FRCPath. Developmental Toxico-Pathologist

#### References

Axelrad JC, Howard CV, McLean WG (2002). Interactions between pesticides and components of pesticide formulations in an *in vitro* neurotoxicity test. *Toxicology*;173(3):259–68.

Axelrad JC, Howard CV, McLean WG (2003). The effects of acute pesticide exposure on neuroblastoma cells chronically exposed to diazinon. *Toxicology*;185(1–2):67–78.

Howard CV (1997). Synergistic effects of chemical mixtures – Can we rely on traditional toxicology? *The Ecologist*;27(5);192–5.

Lang L (1995). Strange brew: Assessing risk of chemical mixtures. *Environ Health Perspect*;103(2):142–5.

Liem AKD, Theelen RMC (1997). *Dioxins: Chemical Analysis, Exposure and Risk Assessment*. PhD Thesis, University of Utrecht, The Netherlands. ISBN 90-393-2012-8.

Vetter W, Luckas B. (1995). Theoretical aspects of polychlorinated bornanes and the composition of toxaphene in technical mixtures and environmental samples. *Sci Total Environ*;160–1:505–10.

Vetter W. Scherer G. (1998). Variety, structures, GC properties, and persistence of compounds of technical toxaphene (CTTs). *Chemosphere*;37(9–12):2525–43.

## **Executive Summary**

In early 2003, Greenpeace exposed the presence of persistent, bioaccumulative chemical pollutants in samples of house dust taken from homes across the UK. Research published by Greenpeace in October 2003 reveals that these same chemicals can be found in many consumer products readily available on every high street.

This report completes the loop of chemical exposure by illustrating two disturbing developments. Firstly, that many of the same chemicals used routinely in consumer products and present in house dust, are also present in the human body, including in prenatal and newborn children. Secondly, that these chemicals are likely to be having a detrimental effect on the health of children and the human population at large. Significantly, the report also draws together the available evidence that illustrates how and why prenatal and newly born children are particularly at risk from chemical pollutants.

The evidence presented here, from academics, governments and well-respected international institutions such as the World Health Organisation (WHO), is not easily dismissed, contributing as it does to a growing bank of international research that reinforces the conclusion of this report – that current chemical legislation is failing to protect children from a harmful chemical assault that begins from a child's conception.

The study focuses on seven key chemicals: alkylphenols (nonyl and octylphenol), bisphenol A, brominated flame retardants, organotins, phthalates, chlorinated paraffins and artificial musks. The report uses available research to demonstrate:

- the presence of these substances in children (and the human population at large)
- the ways in which children are particularly exposed to these substances
- how this increased exposure increases the potential for detrimental health impacts
- the different illnesses and diseases that are now being linked to this chemical exposure
- the specific health impacts of the key chemicals listed above.

Studies have shown that alkylphenols, found in a range of products such as leather and textile finishers and in some detergents, contaminate children before and after birth. Nonylphenol has been detected in umbilical cords, confirming that this hormone-disrupting substance is able to cross the placenta. Preliminary studies show that nonylphenol may disrupt the human immune system by adversely effecting groups of white blood cells.

Likewise, bisphenol A, which is used commonly in electrical products and as laminate coatings for the inside of food tins and bottle tops has been shown to cross the placenta and has been identified in a wide variety of human blood and tissue samples, including foetal tissue. Numerous studies have demonstrated the ability of bisphenol A to alter male reproductive organs and affect behaviour in animals at doses only a little above the amount that human infants have been shown to ingest the chemical.

One group of brominated flame retardants (BFRs) has now been banned in Europe, but others are still in use and many older



products contain the restricted substances. BFRs have been found in human blood, liver and adipose tissue, and in breast milk. Studies have shown that where BFRs are found at high levels in foetal blood, levels tend to be high in the mother's milk and blood. This indicates that maternal levels of BFRs are useful in predicting potential foetal exposure. Some BFRs have been proven to disrupt the genetic processes in cells, which in turn is known to provoke a number of diseases, including cancer.

Research has led to similar findings for organotin compounds, chlorinated paraffins and artificial musk compounds. That research is also described in this report.

Research is now beginning to illustrate that the exposure of prenatal and newborn children to these chemicals is likely to be more significant than the exposure of adults in the same environment. Recent studies have uncovered differences between children and adults in their patterns of absorption, metabolism and excretion of environmental chemicals. Children may absorb chemicals more efficiently, process them more slowly and eliminate them less efficiently than adults.

The final part of this report focuses on the kinds of non-infectious disease, such as cancer, allergies, nervous system disorders, and developmental and reproductive disorders, that are on the rise around the world. A growing body of evidence is pointing to environmental factors as contributors to these problems. Many of the chemicals that have been found in the human body have been shown to be harmful to human and animal health, and we now know that it is children that are particularly at risk. Many of these illnesses can stem from childhood, and many are suspected of being caused by damage to the developing child. This report uses the available research to illustrate the ways in which chemicals are able to damage the immune and nervous systems, interfere with hormonal activity, promote cancer and lead to other detrimental health effects.



### 1 INTRODUCTION

Every year around 100,000 different types of chemicals are produced for a wide range of uses around the world (EEA, 1999). Many of these chemicals find their way into our environment. Other chemicals that have been banned still leak from old products and linger in the environment, and new chemicals are synthesised every year.

Recent governmental and independent reports on the effects of chemicals on the environment and human health still focus predominantly on pollutants about which we already have a significant knowledge, while largely ignoring many other chemicals about which we know little. For example, the recent US Environmental Protection Authority report (US EPA, 2003b) on environmental pollutants and their effects on children, focused on air contaminants such as ozone and sulphur dioxide; the heavy metals lead and mercury; tobacco smoke; and pesticides. This report neglected a vast range of other chemicals which are known to contaminate humans, and about which evidence of harm is building.

The Environmental Working Group (EWG), an independent group in the USA, found 167 chemicals in the blood and urine of nine adult Americans (Houlihan et al., 2003). None of these volunteers worked with chemicals as part of their jobs, and yet each of them contained an average of 91 of the 210 chemicals tested for. Another study released around the same time by the Centers for Disease Control and Prevention (CDC) reviewed 116 chemicals in the blood and urine of adults (CDC, 2003), overlapping the EWG report by 49 chemicals. Both studies

tested for certain classes of chemicals such as polychorinated bisphenols (PCBs), furans, dioxins, organophosphate and organochlorine pesticides and metabolites, heavy metals, and phthalates. The EWG included a much wider range of PCBs and pesticides, and tested for and found dozens of other semivolatile and volatile organic compounds that are found in common consumer products. The CDC included more of the heavy metals, polycyclic aromatic hydrocarbons, well known phytoestrogens, organophosphate pesticides, and other insecticides, herbicides and fungicides.

Most of these chemicals are well-known toxins and have been studied extensively. Many are already banned or have restricted use. Neither the CDC nor the EWG report looked at levels of chemicals in children. Aside from the phthalates, neither report included a range of persistent organic pollutants (POPs) found throughout a large range of everyday household products: the alkylphenols, bisphenol A, brominated flame retardants, chlorinated paraffins, organotins, phthalates, and artificial musks.

These chemicals contaminate our air, water, soil and food. They persist in the environment and accumulate in the tissues of a range of animal life from marine snails to mammals. There is considerable evidence that these chemicals are toxic. Evidence is mounting that tiny levels can effect human health, but the level of contamination in humans is largely unknown.

Limited data shows that these chemicals are found in humans. They accumulate in our



bodies, not just from our day-to-day exposure to various products containing them, but from the time we are conceived and developing in the womb. This is the time when we are most sensitive to chemicals and their toxic effects, and the time at which damage can cause life-long health problems.

There is a wide range of health problems that affect children, or have their origins in childhood, that have been increasing in the last 50 years. These include birth defects, cancer, asthma, immune system disorders, developmental and reproductive disorders, and

nervous system disorders. Many of the POPs that contaminate our bodies have been linked to these diseases.

The following sections review these POPs and their potential health effects. Section 2 reviews the levels of contamination in humans, focusing on contamination in children where data exists. Section 3 reviews how these chemicals find their way into the body and why children are more exposed. Section 4 reviews the health effects of this contamination, focusing on child health. Section 5 provides a summary and conclusion.



## 2 CHEMICAL CONTAMINANTS IN CHILDREN

Accurately estimating the numbers and levels of chemicals that humans are exposed to is almost impossible. It is too difficult to estimate contamination when it comes from so many different sources. Levels of chemicals in female blood can give us an idea of which chemicals an unborn child may be exposed to, but actual levels in the foetus can vary significantly from the mother's levels, depending on how much is absorbed by the placenta. Measures of contamination in the umbilical cord at birth or amniotic fluid, if amniocentesis is performed during pregnancy, are the best indicators of child contamination. Other than levels in blood, levels of chemicals in breast milk and infant food are a good indication of neonatal exposure.

The following section reviews a series of POPs with wide-ranging uses and explores the increasing evidence of their contamination of the child.

#### 2.1 Alkylphenols

Alkylphenols are non-halogenated (halogens include bromine and chlorine) chemicals largely used to make surfactants (detergents), as well as emulsifiers, dispersants and/or wetting agents in industry and consumer applications. Alkylphenols are predominantly manufactured as alkylphenol ethoxylates (APEs), however alkylphenols themselves can be used as plasticisers in plastics, and their derivatives, alkylphenol phosphites, can be used as UV stabilisers in plastics. The most widely used alkylphenols are nonylphenols, followed by octylphenols.

APEs are found in industrial detergents, such as for wool washing and metal finishing; industrial processes such as emulsion polymerisation, leather and textile finishers; laboratory detergents including Triton X-100; some pesticides and other agricultural

products; and the spermicidal lubricant nonoxynol-9. Outside Europe, APEs may also be used in many domestic products, such as liquid clothes detergents in the USA.

In Europe, most APEs have been replaced with safer alcohol ethoxylates. However, in Germany 1,000 tonnes/year of APEs are used in domestic cleaning products despite a voluntary ban on these applications (ENDS, 1999a).

#### 2.1.1 Alkylphenols in the Child

There have been few studies on levels of human contamination by alkylphenols, but those that have been performed clearly show that children are contaminated before and after birth (Table 2.1).

Table 2.1 **Human contamination by nonylphenol.** 

Source	Levels	Study
Umbilical cord	2 ng/kg wet tissue	Takada et al., 1999
Breast milk	0.3 mg/kg lipid	Guenther et al., 2002

Nonylphenol has been detected in human umbilical cords (Takada et al., 1999), confirming that it crosses the placenta from the contaminated mother to the growing foetus. The authors stressed the importance of further studies using larger numbers of umbilical cords, and analyses of maternal and cord blood to estimate the fraction of contaminants passing from the blood to the foetus. Nonylphenol also contaminates breast milk (Guenther et al., 2002).

#### 2.2 Bisphenol A

Bisphenol A is a key building block of polycarbonate plastic and epoxy resins.

Polycarbonate plastic is lightweight, tough, and clear with high heat and electrical resistance, and is therefore used in a wide variety



of products including digital media (e.g. CDs, DVDs), electrical and electronic equipment, automobiles, sports safety equipment, reusable food and drink containers, medical equipment, and many other products. Bisphenol A is also used as an additive in other plastics. Epoxy resins are used in electrical laminates for printed circuit boards, composites, paints and adhesives, and a variety of protective coatings. Cured epoxy resins are used as protective liners in metal cans to maintain the quality of canned foods and beverages, in bottle tops, and in water supply pipes. Some polymers used in dental treatment also contain bisphenol A. In Germany alone, approximately 210,000 tonnes of bisphenol A are produced per year (Leisewitz & Swartz, 1998).

#### 2.2.1 Bisphenol A in the Child

The subject of the effects of bisphenol A on humans has been contentious. It has been suggested that it is only partially absorbed, has a high conversion rate to the biologically inactive bisphenol A monoglucuronide, is rapidly excreted, and shows no evidence of bioaccumulation in tissues (Schonfelder et al., 2002b). For these reasons, until recently, many scientists believed that the active parent form of bisphenol A would not be found in human plasma, and therefore no significant levels could reach the foetus.

However, studies from Germany and Japan have now confirmed that children are exposed to bisphenol A before birth. The first, a Japanese study, found bisphenol A in umbilical cords (Takada et al., 1999). Studies on mice and monkeys then showed that this chemical can cross the placenta (Uchida et al., 2002). At least six studies have since demonstrated bisphenol A in a wide variety of blood and tissues from healthy adults, and maternal and foetal tissues (Table 2.2).

Data from Ikezuki et al. (2002) suggest that bisphenol A may concentrate in amniotic fluid as it was found at approximately 5-fold higher concentrations at 15–18 weeks gestation, compared with other fluids. Schonfelder

Table 2.2 **Human contamination by bisphenol A.** 

maman comamina	on by biophonor / a	
Source	Levels	Study
Umbilical cords	1.6 ng/g wet tissue	Takada et al., 1999
Healthy human blood	0.32 ng/mL	Inoue et al., 2000
Normal adult male blood	1.49 +/-0.11 ng/mL (mean)	Takeuchi & Tsutsumi, 2002
Normal adult female blood	0.64 +/-0.10 ng/mL (mean)	Takeuchi & Tsutsumi, 2002
Blood from women with polycystic ovary syndrome	1.04 +/-0.10 ng/mL (mean)	Takeuchi & Tsutsumi, 2002
Ovarian follicular fluids obtained during in vitro fertilisation procedures	1–2 ng/mL (mean)	lkezuki et al., 2002
Blood from perimenopausal and pregnant women	1-2 ng/mL (mean)	lkezuki et al., 2002
Amnoitic fluid at 15–18 weeks	8.3 +/-8.7 ng/mL (mean)	lkezuki et al., 2002
Maternal blood at 32-41 week gestation	3.1 ng/mL (median ) 0.3–18.9 ng/mL	Schonfelder et al., 2002b
Umbilical cord blood	2.9 ng/mL (median) 0.2–9.2 ng/mL	Schonfelder et al., 2002b
Placental tissue	12.7 ng/g (median) 1.0–104.9 ng/g	Schonfelder et al., 2002b
Amniotic fluid, normal karyotype	0.26 ng/mL (median) 8 samples showed levels of 2.80–5.62 ng/mL	Yamada et al., 2002
Amniotic fluid, abnomal foetal karotype	0 ng/mL (median)	Yamada et al., 2002
Maternal blood, normal foetal karyotype	2.24 ng/mL (median)	Yamada et al., 2002
Maternal blood, abnormal foetal karyotype	2.97 ng/mL (median)	Yamada et al., 2002
Maternal blood	0.46 +/-0.20 ng/mL (mean)	Kuroda et al., 2003
Umbilical cord blood	0.62 +/-0.13 ng/mL (mean)	Kuroda et al., 2003
Sterile patients' blood	0.46 +/-0.20 ng/mL (mean)	Kuroda et al., 2003
Sterile patients' ascitic fluid	0.56 +/-0.19 ng/mL (mean)	Kuroda et al., 2003



et al. (2002b) also showed that in 14 of 37 cases, foetal plasma levels of bisphenol A were higher than in the corresponding mother's blood. Foetal levels of bisphenol A were also significantly higher in males, which may indicate sex differences in the metabolism of this chemical. Takeuchi and Tsutsumi (2002) confirmed this gender difference in a study on adults, and suggested that it may be due to androgen (male hormones) related metabolism of bisphenol A.

#### 2.3 Brominated Flame Retardants

Brominated flame retardants are a diverse group of compounds used to prevent combustion and/or retard the spread of flames in a variety of plastics, textiles and other materials. They can be found in a large range of industrial and electrical appliances, vehicles, lighting and wiring, textiles including carpets and furnishings, and in packaging and insulating materials such as polystyrene. Over 70 compounds or groups have been described, but three dominate:

- polybrominated diphenyl ethers (PBDEs)
- hexabromocyclododecane (HBCD)
- brominated bisphenols (such as TBBP-A)

PBDEs and HBCD are mainly used as additives. TBBP-A is more commonly used as a reactive component, becoming more tightly bound to the polymers in which it is incorporated, but some additive uses still exist. Polybrominated biphenols (PBBs) are no longer produced in Europe but substantial quantities remain in existing and imported products and wastes.

One group of PBDEs, the penta-BDEs, are being banned across Europe, however, they may still appear in the environment following degradation. For example, UV light and sunlight can cause deca-BDE to debrominate, resulting in the formation of lower-brominated diphenyl ethers such as penta-BDEs (Darnerud et al., 2001). The World Health

Organisation has recommended that PDBEs not be used where suitable replacements are available, and Swedish and Danish Governments have called for PDBEs to be phased out. PBDEs were produced at a rate of 40,000 metric tons per year world-wide in 1999 (Rahman et al., 2001).

## 2.3.1 Brominated Flame Retardants in the Child

The majority of the contamination data is for PBB (now largely disused), and the PBDEs. Levels of human contamination by other brominated flame retardants is largely unknown.

The similarity of PBDEs to dioxins and polychlorinated biphenyls (PCBs) has been of major concern because their toxic effects may be similar, therefore these are the most studied group of brominated flame retardants. PDBEs are found in human breast milk, blood, liver and adipose tissue (Table 2.3). Extensive breast milk studies in Sweden show an exponential increase in PBDEs in breast milk (an average increase from 0.07-4.02 ng/g lipid between 1972-1997) while other chemical contaminants such as PCBs have tended to decrease (Meironyte et al., 1999). However, a recent paper has reported a decrease of PBDEs in Swedish breast milk since 1997, possibly due to a voluntary phase out of penta-BDE (Hooper & She, 2003).

This rise in human contamination was also seen in a study on human blood samples (10 male, 10 female) taken in 1985–1999 for the German environmental specimen bank. The median PBDE concentrations in blood have increased from 3.1–4.7 ng/g lipid (Schroter-Kermani, 2001). This study also showed that males may be more at risk of contamination than females as male blood samples showed 20–75% higher levels of PBDEs than female blood, depending on the sampling year.

Babies born to mothers in the USA may be most at risk of PBDE contamination. Mazdai



Table 2.3 Human contamination by polybrominated diphenyl ethers.

PBDE	Source	Levels	Study
TBDE	Human adipose tissue – healthy subjects	5.1 ng/g (mean) 0.6–27.5 ng/g	Hardell et al.,1998
TBDE	Human adipose tissue – non-Hodgkin's lymphoma patients	13.0 ng/g (mean) 1.0–98.2 ng/g	Hardell et al.,1998
PBDEs (8 identified including BDE-47)	Breast milk	4.02 ng/g lipid (mean)	Meironyte et al., 1999
PBDEs (BDE-28, 47, 66, 85, 99, 100, 153, 154)	Human blood (10 male, 10 female)	4.7 ng/g lipid (mean)	Schroter-Kermani, 2001
PBDEs (BDE-47, 99, 153, 154, 100)	Breast adipose	86 ng/g (mean)	She et al., 2002
PBDEs	Breast milk	0.67-2.84 ng/g lipid	Ohta et al., 2002
PBDEs (BDE-28, 47, 99, 100, 153, 154, 183)	Human adipose tissue	1.29 ng/g lipid (median) 0.47–2.75 ng/g lipid	Choi et al., 2003
PBDEs	Maternal blood	24.0 pg/g fresh weight (median)	Guvenius et al., 2003
PBDEs	Umbilical cord blood	4.3 pg/g fresh weight (median)	Guvenius et al., 2003
PBDEs	Breast milk	75.0 pg/g fresh weight (median)	Guvenius et al., 2003
PBDEs (6 including BDE-47)	Maternal blood	15–580 ng/g lipid	Mazdai et al., 2003
PBDEs (6 including BDE-47)	Umbilical cord blood	14–460 ng/g lipid	Mazdai et al., 2003

PBDEs: polybrominated diphenyl ethers; TBDE: 2,2',4,4'-tetrabrominated diphenyl ether

et al., (2003) found that the concentrations of PBDEs in maternal and foetal serum samples in Indianapolis, IN, USA, were 20–106 times higher than the levels reported previously in Swedish mothers and infants (Guvenius et al., 2003) and 20 times higher than Norwegian blood samples (Thomsen et al., 2002). In San Francisco Bay, CA, USA, breast adipose tissue samples from 23 women are the highest human levels reported to date (She et al., 2002).

There is a high correlation between levels of PBDE in foetal blood and the corresponding levels in the mother's milk and blood, indicating that maternal PBDE blood levels are useful in predicting foetal exposure (Mazdai et al.,2003)

The tetrabrominated PBDE congener, BDE-47, is the most abundant PBDE congener in

all human samples tested, making up 53–65% of total PBDEs detected. The other major congeners include, BDE-99, 100, 153, and 154 (Schroter-Kermani, 2001; She et al., 2002; Mazdai et al., 2003).

#### 2.4 Organotins

Organotins are organic compounds containing at least one bond between carbon and metal tin and include:

- monodibutyltins (MBT)
- dibutyltins (DBT)
- tributyltins (TBT)
- triethyltins (TET)
- triphenyltins (TPT)
- octyltins (MOT, DOT).

Mono-substituted organotins tend to be used as stabilisers, di-substituted organotins



in PVC and as catalysts in production of polyurethane foams and silicones, and trisubstituted organotins in a wide range of agricultural and general biocides, but there is some overlap in uses between the groups. Organotins can therefore be found in: antifoulants on boats, ships, quays, buoys, crab-pots, fishing nets and cages; biocides for cooling systems, power stations, pulp and paper mills, breweries, leather processing and textile mills; slimicides on masonry; wood preservatives; anti-fungal agents in carpets and PVC flooring; glass coating applications; disinfectants; heat stabilisers in rigid (pipes, panels) and soft (wall-coverings, furnishing, toys) PVC products; paint and rubber; pesticides targeting everything from insects, to worms, to rats; and anti-bacterial shoe insoles and nappies.

TBT has been banned for use on small vessels in many countries for over 10 years due to its impact on marine molluscs, and is currently being phased out of larger vessels worldwide (Champ, 2000). PVC uses about two-thirds of the global consumption of the organotins, which can make up to 2% of the finished product. Approximately 15,000 tonnes of organotins were used in PVC in Europe in 1995 (Santillo et al., 2003).

#### 2.4.1 Organotins in the Child

Although organotins, particularly TBT, have been found in a wide range of molluscs, fish, marine birds, marine mammals, and freshwater birds (IPCS, 1999), aside from a few reports (Table 2.4), levels of contamination in humans are largely unknown, and there are no readily available reports on child contamination.

#### 2.5 Phthalates

Phthalates are among the most ubiquitous synthetic chemicals in the environment. They are non-halogenated ester derivatives of phthalic acid. Some phthalates are used as discrete products, such as di(ethylhexyl) phthalate (DEHP), while others are complex

isomeric mixtures containing individual compounds with similar structures such as the di-iso-nonylphthalates (DINP) and di-iso-decylphthalates (DIDP).

The major use for phthalates are as plasticisers in PVC products including building/furnishing materials, floorings, furniture, food packaging, toys, clothing, car interiors, cables, and a range of medical equipment such as blood bags. They are also used in solvents, lubricating oils, fixatives, adhesives, paints, sealants, surface coatings, insecticides, detergents, printing inks, car-care products, soaps, shampoos, hand lotion, nail polish, cosmetics, and perfumes. Six phthalates have been banned from use in toys designed to be put into the mouths of children under 3-years old.

Over 1 million tonnes of phthalates are produced each year in Europe alone, primarily for use in the EU (Santillo et al., 2003). The main phthalates are DEHP, DINP, DIDP, dimethylphthalate (DMP), diethylphthalate (DEP), and dibutylphthalate (DBP). DEHP constitutes about 50% of the market for PVC plasticisers in Western Europe (EC, 2003).

#### 2.5.1 Phthalates in the Child

Metabolites of phthalates in the urine indicate a wide exposure of humans to phthalates (Barr et al., 2003; CDC, 2003; Koch et al., 2003). Phthalates have also been detected in adult blood (see Table 2.5), but child studies are scarce.

A study on premature breast development (thelarche) in female children aged 6 months to 8 years, found phthalate esters in 68% of serum samples from the thelarche patients (Table 2.6). The phthalate esters with the most common commercial uses, DEHP and DBP, were detected in the highest concentrations. For those samples with high concentrations of DEHP, one of the major DEHP metabolites, mono(2-ethylhexyl)phthalate (MEHP), was also detected. DEHP was detected in only 14% of the control samples in lower



Table 2.4 **Human contamination by organotins.** 

Organotin	Source	Levels	Study
BTs	Human male livers	59-96 ng/g (79% TBT)	Takahashi et al., 1999
TPT	Human blood	0.17-0.67 mg/L	Lo et al., 2003

BTs: butyltins; TBT: tributyltin; TPT: triphenyltin.

Table 2.5 Human contamination of blood from adults by the phthalate DEHP.

Group	Range	Median	Positive Samples (%)	Study
Endometriosis patients (n=55)	0–3.24 μg/mL	0.57 μg/mL	92.6	Corbellis et al., 2003
Normal females (n=24)	0–1.03 μg/mL	0.18 μg/mL	-	Corbellis et al., 2003
Normal adults (6 female, 3 male)	97.2–904.8 μg/g lipid	-	100*	Houlihan et al., 2003

DEHP: di(ethylhexyl) phthalate.

Table 2.6

Contamination of blood from normal girls and girls with premature breast development (thelarche) by phthalates.

Phthalate	Group	Positive Samples	Range of positives (mg/mL)	Group Mean (mg/mL)
BBP	Thelarche	2/41	0.054–0.117	0.004
BBP	Normal	0/35	0	0
DBP	Thelarche	13/41	0.015–0.276	0.042
DBP	Normal	0/35	0	0
DEP	Thelarche	3/8	0.0019-0.0089	0.0055
DEP	Normal	nd	nd	nd
DEHP	Thelarche	25/41	0.187–2.098	0.450
DEHP	Normal	5/35	0.276–0.719	0.070
DEP	Thelarche	5/41	0.008-0.037	0.003
DEP	Normal	0/35	0	0
DMP	Thelarche	2/8	0.0183-0.0214	0.01985
DMP	Normal	nd	nd	nd
DOP	Thelarche	1/41	0.438	0.011
DOP	Normal	1/35	0.562	0.016
MEHP	Thelarche	5/41	0.0063-0.038	0.003
MEHP	Normal	0/35	0	0

BBP: benzylbutylphthalate; DBP: dibutylphthalate; DEHP: di-(ethylhexyl)phthalate; DEP: diethylphthalate; DMP: dimethylphthalate; DOP: dioctylphthalate; MEHP: mono(ethylhexyl)phthalate. (Data from Colon et al., 2000).

<sup>\*6</sup> phthalates were detected in all 9 subjects but only DEHP was quantified.



concentrations. A more sensitive analysis of eight thelarche samples allowed detection of a further two phthalates, DMP and DEP, in two and three samples, respectively.

While there are few studies on the contamination of young children, other studies suggest that their levels could be at least as high as adult levels. A large study of urine phthalate levels in 2,541 USA residents showed that females tended to have higher concentrations than males (CDC, 2003). Animal studies show that phthalates cross the placenta and pass into breast milk (Dostal et al., 1987; Parmar et al., 1985; Srivasta et al., 1989), therefore, phthalates will be passed onto developing

foetuses and newborn children via their mothers. Additionally, children seem to be more exposed to phthalates than adults. In the CDC study, of the seven urine phthalate metabolites tested, the highest levels of metabolites for DEHP, DBP and monobenzylphthalates were found in the youngest agegroup tested: the 6–11-year-old children (CDC, 2003).

#### 2.6 Artificial Musks

Nitro musks and polycyclic musks are a series of structurally similar aromatic compounds used as substitutes for natural musks. Artificial musks are cheap, easy to

Table 2.7 Human contamination by artifical musks.

Musk	Sample	Levels	Positive Samples	Study
Xylene	Breast milk (n=391)	100 μg/kg fat (mean) 10–1220 mg/kg	-	Liebl & Ehrenstorfer, 1993
Ketone	Breast milk (n=391)	40 μg/kg fat (mean)	-	Liebl & Ehrenstorfer, 1993
Galaxolide	Adult adipose (n=14)	16-189 mg/kg fat	-	Rimkus & Wolf, 1996
Tonalide	Breast milk (n=5)	8-58 µg/kg fat	-	Rimkus & Wolf, 1996
Xylene	Adult blood in 1993 (n=72)	0.24 ~µg/L (median) <0.10-1.12 mg/L	66/72	Angerer & Kafferlein, 1997
Xylene	Adult blood in 1998 (n=41)	<0.1 µg/L (median) <0.1–0.29 mg/L	5/41	Kafferlein & Angerer, 2001
Xylene	Blood from women with endocrine and gynaecological problems (n=154)	65.5 ng/L (median) (1183 ng/L maximum)	95%	Eisenhardt et al., 2001
Ketone	Blood from women with endocrine and gynaecological problems (n=154)	55.5 ng/L (median) 518 ng/L (maximum)	85%	Eisenhardt et al., 2001
Mosken	Breast milk (n=53)	64 μg/kg fat (mean)	4/53	Zehringer & Herrmann, 2001
Tibeten	Breast milk (n=53)	25 μg/kg fat	1/53	Zehringer & Herrmann, 2001
Xylene	Breast milk (n=53)	35 μg/kg fat	1/53	Zehringer & Herrmann, 2001
HHCB	Breast milk (n=53)	73 µg/kg fat (mean)	52/53	Zehringer & Herrmann, 2001
AHTN	Breast milk (n=53)	44 μg/kg fat (mean)	37/53	Zehringer & Herrmann, 2001
Traseolide	Breast milk (n=53)	74 μg/kg fat	1/53	Zehringer & Herrmann, 2001



produce fragrances that are added to personal care and household products such as laundry detergents, shower gels, soaps, hand lotions and perfumes. Some musks are also used as additives in food, room fragrances, chewing tobacco, fish baits, Indian joss sticks, and in technical products such as herbicide formulations and explosives. The nitro musks include musk xylene, musk ketone, musk ambrette, musk tibetene and musk moskene. The major polycyclic musks are tonalide and galaxolide, also know as AHTN and HHCB respectively.

Musk ambrette has been banned and musk xylene restricted due to health concerns. They are largely being replaced with polycyclic musks which may appear less problematic but are, in fact, largely untested.

#### 2.6.1 Artificial Musks in the Child

Nitro musks have been found in human adipose, breast tissue and blood (Table 2.7) but no studies appear to have been done on children. However, adult female blood and breast milk levels indicate that contamination of the foetus and newborn does occur.

#### 2.7 Chlorinated Paraffins

Polychlorinated-n-alkanes or chlorinated paraffins (CPs) are produced by reacting chlorine gas with paraffins (hydrocarbons) (Fig. 2.6). They are complex mixtures and exist as oils or solids depending on their content of chlorine and carbon. They are divided in to six groups depending on carbon chain length (short C10–13; intermediate C14–17; long C18–30) and degree of chlorination (low <50%; high >50%).

They are used as high-temperature lubricants, plasticisers, flame retardants, industrial cutting oils for metal working, finishing agents for leather goods and textiles, and additives in adhesives, paints, rubber and sealants. Some short chain CPs are being used as replacements for PCBs. They are also used as secondary plasticisers for PVC or as partial replacements for primary plasticisers such as phthalates. In 1998, about 4000 tonnes of CPs were produced in the EU (Santillo et al., 2003).

#### 2.7.1 Chlorinated Paraffins in the Child

Data on the human contamination by CPs is scarce. One report from 1985 on the contamination of environmental samples by CPs includes a sample of human fat (200 µg/kg fat) (Schmid & Muller, 1985).



## 3 CHILD EXPOSURE TO CHEMICALS

The suite of chemicals to which children are exposed is not measured or defined by any programme. Testing, labelling and monitoring programmes capture only a fraction of the chemicals we are exposed to. Children are exposed to so many different sources of chemicals that it is very difficult to estimate which chemicals are important and at what levels. In addition, due to the combined effects of development, physiology, diet and behaviour specific to their age group, children may experience higher levels of exposure to toxins than adults within the same environment.

An increasing number of studies have uncovered differences between children and adults in their patterns of absorption, metabolism and excretion of environmental chemicals. Children may absorb chemicals more efficiently, process them more slowly and eliminate them less efficiently than adults. Children may be more or less sensitive to the particular effects of a chemical, and the chemical may produce different adverse effects. It is clear that studies of adults must be used cautiously to predict potential adverse effects in children.

This section looks at the ways in which children are contaminated by chemicals, and explores why they are more susceptible to this contamination.

#### 3.1 From Products to Children

About 100,000 different chemicals are registered for commercial use within Europe with hundreds of new chemicals registered each year (Goldman, 2002). In 2001, major USA industries reported the release of 6.16 billion pounds of industrial chemicals into air, water and landfill (US EPA, 2003b), just a portion of what is actually released. Chemicals can also leach from their products, contaminating food, homes and work places.

The majority of the chemicals described in Section 2 are persistent; that is, they stay in the environment for a long time. Those chemicals that are described as having a relatively short half-life are regularly released into the environment in such large amounts that they are constantly detectable. For example, despite having a relatively short half-life in the body (about 12 hours), phthalate metabolites in the urine of women in one study remained fairly constant from day to day, suggesting a constant daily exposure (Hoppin et al., 2002).

Many of these chemicals are also bioaccumulative; that is, they accumulate in the bodies of organisms, some in the fat tissue and others in specific sites such as liver and kidney. These chemicals then build up in the food chain so that those organisms at the top of food chains, such as humans, have the highest exposure. The developing child, who feeds directly from the mother's body or breast milk, sits at the top of many food chains.

#### 3.1.1 Prenatal Exposure

By the third month of pregnancy, the placenta has sent extensive branches into the deep layers of the uterus (Moore & Persaud, 2003). These are intertwined with the spiral arteries of the uterus which send jets of blood between these branches. As the mother's blood trickles between the placental branches, carbon dioxide and metabolic wastes are taken away and replaced with oxygen, water, minerals, nutrients and antibodies. This is a similar process by which the blood in our capillaries are cleaned and refuelled, with one major difference – the placenta actively pumps most of what it needs (with the exception of oxygen) from the mother rather than relying on passive diffusion. Therefore, the foetus is guaranteed a constant supply of required substances, even if the mother's blood levels are unusually low or high in a particular component. In this way, the placental barrier can decide, to some extent, what enters the developing child.



When it comes to toxic chemicals that have contaminated the mother, however, the placenta is not really a barrier. Chemical substances from the mother's blood are sorted primarily on the basis of molecular weight, electrical charge and lipid solubility. Small, neutrally charged molecules that readily dissolve in fat are given free passage through the placenta, regardless of their potential for harm. Methylmercury, for example, is actively pumped from the mother's blood so that the mercury levels in the umbilical cord will eventually surpass the levels in the mother's blood.

The foetus, suspended by the umbilical cord, floats freely in amniotic fluid which surrounds, protects, and bathes the developing foetus (Moore & Persaud, 2003). This fluid, made by tissues from both the foetus and mother, is constantly sipped and swallowed by the foetus, and is absorbed by the digestive tract. It soaks through the skin, and is inhaled as the baby practices breathing. It is absorbed by the foetus, enters the foetal circulation, and leaves via the maternal circulation. It bathes the inside and outside of the developing baby, as do any chemicals which contaminate it.

As well as being exposed to the chemicals which the mother is exposed to daily, the developing foetus is exposed to chemicals that have been stored in her tissues, and are released during pregnancy. For example, there is a dramatic mobilisation of maternal

fat stores during the third trimester, a period critical to brain development (Moore & Persaud, 2003).

Bisphenol A is a good example of the problem. In many cases, foetal plasma levels of bisphenol A are higher than for maternal blood (Schonfelder et al., 2002b). Bisphenol A concentrations in amniotic fluid at 15–18 weeks gestation were five times higher than in other fluid in one study (Ikezuki et al., 2002). The rate of clearance of bisphenol A from the blood is slower in foetuses because the enzymes required to clear it are not expressed until after birth. Women also have higher concentrations in their blood than men, due to differences in either exposure or metabolism between the sexes (Schonfelder et al., 2002).

#### 3.1.2 Postnatal Exposure

#### Food

The major source of exposure to bioaccumulative chemicals is food (for example, see Table 3.1). Food is first contaminated via environmental pollution during growth and bioaccumulation along the food chain. This contamination continues due to leaching from components and packaging used during manufacturing, processing and storage, particularly for food containing higher levels of fat.

Fish tends to be the most highly contaminated of all food. For example, the flame retardant PBDEs have been detected in fish and

Table 3.1 A comparison of the estimated daily intake of the phthalate, DEHP, from varying sources by different age groups in Canada (μg/kg body weight per day).

Age (years)					
Source	0.0-0.5	0.5–4	5–11	12–19	20–70
Food	7.9	18	13	7.2	4.9
Indoor air	0.86	0.99	1.2	0.95	0.85
Drinking water	0.13-0.38	0.06-0.18	0.03-0.10	0.02-0.07	0.02-0.06
Soil	0.000064	0.000042	0.000014	0.000004	0.00003
Ambient air (Great Lakes)	0.00003-0.0003	0.00003-0.0003	0.00004-0.0004	0.00003-0.0003	0.00003-0.0003
Total	8.9–9.1	19	14	8.2	5.8

(Data from Meek & Chan, 1994).



shellfish in the range of 21–1650 pg/g fresh weight (Ohta et al., 2002). In comparison, beef, pork, and chicken contained 6.25–63.6 pg/g, and three different vegetables had levels of 38.4–134 pg/g. Similarly, in Japan where fish is a major part of the diet, the estimated daily intake of the organotin TBT is relatively high (2.5 µg/kg body weight based on a fish consumption of 150 g/day) (van Heijst, 1994). Ohta et al. (2002) also showed a strong correlation between PBDE levels in breast milk and intake of fish and shellfish.

Preparation and storage of food can increase contamination dramatically. Bisphenol A, which leaches from unreacted materials and/or degraded polymers from plastics, is a good example. Bisphenol A levels migrating from plastic products into baby food increase after dishwashing, boiling, and brushing (Brede, 2003); it migrates from rubber products and plastic stretch-film used in food-contact (Lopez-Cervantes & Paseiro-Losada, 2003; Ozaki & Baba, 2003); and an estimated 80–85% of food cans contain bisphenol A.

In the case of the organotins, TBT, DBT, and MBT can leach from baking parchment, while DBT and MBT leach from gloves for kitchen work, dish-washing sponges, and cellophane film for food (Takahashi et al., 1999). Similarly, in another study, the process of frying and packaging chicken increased the phthalate DEHP content from 0.08–16.9 mg/kg. The main source of contamination was PVC gloves used by food workers (Tsumura et al., 2001b). PVC tubing has also been a source of high-level contamination of baby food (Tsumura et al., 2001a)

As an example of the extent of chemical contamination of food, a recent report found nonylphenol in all 60 food-products tested (Guenther et al., 2002). These were all popular, common foods in Germany and included 39 adult foods, from marmalade to sausages, 20 baby foods, and one sample of breast milk (see Table 3.2 for a selection of

the results). Nonylphenol was detected in every sample within the range of  $0.1\text{--}19.4~\mu\text{g/kg}$ , and was not related to fat content. The authors pointed out that the foods varied substantially in preparation methods and packaging and that there are multiple entry points for nonylphenol into the food supply.

Table 3.2 Examples of nonylphenol levels in infant foods, and general foods that may be fed to infants.

Food	Nonylphenol levels (µg/kg)
Breast milk (35-year old mother)	0.3
Infant formula 1	1.6
Infant formula 2	2.1
Banana & milk puree	0.2
Peach & honey puree	0.4
Carrots puree	0.8
Semolina & vanilla puree	1.8
Broccoli, potato, turkey puree	1.4
Beef, potato & rice puree	3.1
Noodles in ham & tomato sauce	4.0
Peach & passion fruit yoghurt	0.6
Whole milk (3.5%)	0.4
Evaporated milk (10%)	3.8
Hen's egg	1.5
Tuna	8.1
Apples	19.4
Orange juice	0.1
(Data from Guenther et al., 2002).	U.1

#### Air

The second highest source of chemical contamination is air (Table 3.1). Indoor exposures to chemicals, including alkylphenols, bisphenol A and phthalates, in preschool children have been shown to be greater indoors than out (Wilson et al., 2003). Chemicals build up from cleaning compounds, personal care products and cosmetics. Vapours and degraded materials leach from carpeting, paints, computers, furniture and toys, and adhere to aerosol particles.



For example, DINP and DEHP are the predominant phthalates that migrate from children's toys and childcare articles (Bouma & Schakel, 2002). Various studies have found a wide range of chemicals in the dust of many houses, including alkylphenols, bisphenol A, organotins, flame retardants, phthalates, and chlorinate paraffins (Santillo et al., 2003).

Outside air contains lower levels of these chemicals, some of which may come from vaporisation from contaminated water and may show seasonal trends (Ying et al., 2002). For example, air in urban New York and New Jersey in the USA, contained nonylphenol at levels of 2.0–70 ng/m<sup>3</sup> due to vaporisation from the Hudson River (Dachs et al., 1999).

#### Water

Water is contaminated with chemicals via sewage, industry waste-water, and leaching from land-fill. APEs in treated waste-water effluents, for example, are similar in Spain, the UK and the USA with concentrations up to 369 mg/L while in sediment they reach up to 13,700 mg/kg in the USA (Ying et al. 2002).

Research by the UK Government's Drinking Water Inspectorate has shown that various products for drinking water pipes (plastic resins, rubbers, cements and pipe-linings) leach alkylphenols and phthalates into the drinking water (ENDS, 1999b). One epoxy resin coating leached two phthalates, DBP and DBEP, at levels of 1,400 µg/m² and 280 µg/m² respectively, and the alkylphenol, nonylphenol at levels of 160 µg/m² into water for the first hour after curing. A solvent-based cement coating leached 12,000 µg/m²/h of nonylphenol polyethoxylate immediately after curing, and leaching rates remained high after three days.

Contamination can even come from water filters. For example, following purification of tap water for laboratory use, bisphenol A levels increased from 0.01–0.02 ng/mL (Inoue et al., 2000).

#### **Medical Sources**

Many medical devices contain plastics, including blood bags, mechanical ventilators, haemodialysis machines, feeding equipment, and infusion lines. Therefore, medical interventions can increase exposure to chemicals used in plastics. Neonates, particularly premature neonates receiving multiple interventions in intensive care units, are exposed to high levels of plastic additives. A particular problem is DEHP, the only phthalate currently used in medical devices, which contain on average 20-40% DEHP by weight (EC, 2002). Bisphenol A can also be a problem as, for example, haemodialysis patients have additional exposure to the chemical through polycarbonate components in dialysis machines (Yamasaki et al., 2001).

#### 3.2 Increased Exposure in Children

#### 3.2.1 Diet

Young children eat 3–4 times more food in proportion to their body size than adults (Table 3.3), and therefore ingest more chemicals per unit of body mass. They also eat a more limited range of food – human milk or cow milk-based products are the predominant source of energy and nutrients in the first year of life. Children are therefore exposed to a higher proportion of lipid-soluble chemicals that are released into milk from the fat stores of both human mothers and cows.

Children also drink more water relative to their body mass. Infants below 6 months old consume 88 mL/kg/day of tap water, directly and via sources such as fruit juice and formula, and infants aged 6–12 months ingest 56 mL/kg/day. In contrast, adults over 20 years ingest only 15 mL/kg/day (Altshuler et al., 2003c).

#### 3.2.2 Behaviour

Ingestion of toxins from non-dietary sources is relatively high in young children because they constantly place objects and fingers in



their mouths, particularly those going through the period characterised by intense oral exploratory behaviour. Soil may be ingested inadvertently or by deliberate or compulsive behaviour, and this may be greater in retarded children (Altshuler et al., 2003). Children also tend to paddle or swim more often, and are thus exposed to more water-borne chemicals.

Table 3.3 Energy intake of children at selected ages (kcal/kg/day).

Age	Males	Females			
0–1 month	116	116			
2-3 months	100	100			
6-9 months	85	85			
2–3 years	81	81			
4–6 years	72	64			
12-13 years	59	52			
18–19 years	48	41			

(Data from Bearer, 1995)

#### 3.2.3 Stature

During infancy and toddler years, children spend a large amount of time on or near the floor or ground. Here they are more exposed to dense vapours, car exhausts, house dust and chemicals leaching from floor coverings.

## 3.2.4 Enhanced Absorption Dermal Absorption

Children have a higher skin surface area to body weight ratio than adults, and experience more intensive contact with home surroundings, so increased dermal absorption of chemicals may occur. The skin of children is also more permeable than adult skin. In newborns, keratinisation (thickening and toughening of the skin) does not occur until 3–5 days after birth, and is more delayed in premature infants (Bearer, 1995). Studies have shown enhanced absorption of toxins including various dyes, drugs and disinfectants through the skin of newborns (Eichenfield & Hardaway, 1999).

#### **Gastrointestinal Absorption**

Gastrointestinal (GI) absorption is increased in children under 6–8months old, and the immature GI tract shows prolonged gastric emptying time and a prolonged intestinal tract transit time. This means that children will be more exposed to ingested chemicals. For example, children absorb 50% of ingested lead compared to 10% by adults. Newborns do not achieve adult levels of stomach acidity until several months of age which may increase or decrease the absorption of different chemicals (Bearer, 1995).

#### **Respiratory Absorption**

Young children breathe about twice as much air in proportion to their body mass as adults and therefore inhale proportionately more airborne chemicals (Altshuler et al., 2003c).

#### **Brain Absorption**

The blood-brain barrier (BBB) limits the ability of most chemicals to penetrate the brain from the general circulation. Lipid-soluble compounds can cross more easily as the lipid content of the brain is high. However, the effectiveness of the BBB is age-dependant – the BBB develops gradually during foetal growth and infancy, and matures during childhood (Altshuler et al., 2003c). This higher permeability to chemicals may result in higher exposures of the brain to toxins.

#### 3.2.5 Distribution in the Body

Blood and body composition and organ size in infants and children differ from adults. This affects the distribution and storage of chemicals, and thus the exposure of organs and tissues to their toxic effects. In the blood, for example, chemicals may be bound to special blood proteins that prevent them from accessing target sites in organs. Bound chemicals are less likely to cause adverse effects in the body. However, serum albumin, the predominant binding protein in blood, only reaches adult levels by the age of 10–12 months (Altshuler et al., 2003c). Thus, infants



are more vulnerable to toxins as they lack this protection. Infants also have a higher percentage of water and a greater volume of extracellular fluid than adults – 75% of body weight at birth compared to 50–60% in an adult – and therefore can carry proportionately more chemicals in their body fluids.

In older children and adults, fat can act as a buffer to lipid-soluble toxins, by storing them and releasing them slowly over a period of time. The developing foetus lacks this buffer as it has little body fat until the last four weeks of gestation (Moore & Persaud, 2003). Fat-soluble chemicals are, therefore, more likely to be stored in foetal tissues with a higher fat content, such as the highly sensitive, developing brain.

Some organs in the child are larger than those of an adult in relation to their body mass, and blood flow to their organs may also be greater in proportion to body mass. For example, the infant brain makes up 13% of total body mass, while an adult brain is only 2%, and cerebral flow is greater per unit mass of brain weight than for an adult (Altshuler et al., 2003c).

All these factors may result in a greater distribution and storage of certain chemicals in the organs of children. An increase in storage can then extend the duration of exposure to chemicals, as they are slowly re-released over time.

#### 3.2.6 Metabolism

Children have a large skin surface in proportion to their body mass and therefore lose body heat more rapidly. This requires a higher resting metabolic rate than adults and more oxygen relative to their body mass. They also tend to be more active than adults, further increasing their respiration rate and energy metabolism. Therefore, both their food and air intake is higher and their exposure to polluting chemicals is greater per unit of body mass than for adults.

#### 3.2.7 Biotransformation of Chemicals

Biotransformation of a chemical into its metabolite(s) may either increase or decrease its toxicity, and make it easier or harder to eliminate it from the body. In infants and children, many important biotransformation pathways do not function at adult levels, and some transformation processes follow different pathways (Bearer, 1995; McCarthy, 2003). These differences can result in an inability to metabolise a chemical, or the production of different metabolites in the developing foetus or child compared to adults. Immaturity can be an advantage for those chemicals where the metabolite is more toxic than the parent chemical; however, in general, reduced metabolic capacity of infants and children probably predisposes them to the risk of more severe adverse effects than adults.

#### 3.2.8 Excretion and Elimination

Chemicals may be excreted as the parent compound, metabolite, or linked to other chemicals (conjugates) via urine, faeces or expelled air. Some chemicals are transformed to water-soluble products so that they can be excreted by the kidneys via glomerular filtration, tubular diffusion, or tubular secretion. Glomerular filtration removes most toxins unless they are tightly bound to plasma proteins.

The liver and kidneys of newborns are not fully developed at birth and so toxins are eliminated more slowly than they are in adults (McCarthy, 2003). The glomerular filtration rate of a full-term newborn is about 50% that of an adult and reaches the full capacity at about 12 months. Tubular secretion and renal blood flow reach adult rates by about 6–9 months and 5–12 months, respectively (McCarthy, 2003).

Due to a combination of differences in biotransformation and immature organ function, it can take neonates 2–9 times longer on average to eliminate a chemical from their body, depending on the primary



## 4 CONTAMINATING CHEMICALS AND DISEASE

A range of non-infectious diseases around the world are on the rise. Environmental factors are believed to contribute to all these diseases, although specific factors or chemicals may not have been identified. All these diseases can stem from childhood, and many are suspected to be caused by damage to the developing child, the period of life most susceptible to chemical damage.

Many of the hundreds of chemicals which have been shown to contaminate our bodies are toxic to a range of animals. The effects of these chemicals on human health are largely unexplored, partly because they are found in doses in the environment that were once considered unlikely to have an effect. Research is now linking low-dose chemical studies in animals to a broad range of health effects that were previously unexplored in high-dose studies. These health effects are all related to diseases seen in humans.

This section reviews the diseases that are on the rise, and describes the periods of child development that are vulnerable to chemical damage. The final part presents the current evidence of the toxic effects of low doses of persistent organic pollutants and their possible links to disease.

#### 4.1 Diseases on the Rise

#### 4.1.1 Infant Mortality

The USA still has one of the highest infant mortality rates and the highest rates of preterm and low birth-weight babies in the industrialised world (Altshuler et al., 2003a). Table 4.1 shows the ten leading causes of death for newborns in the USA. About 55% of these infant deaths are attributed to:

- congenital abnormalities
- complications of low birth weight and prematurity
- Sudden infant death syndrome
- complications of labour and delivery.

The risks of developing these disorders are all increased by exposure to chemicals. While deaths from most of these causes are declining, congenital abnormalities do not appear to be declining.

#### 4.1.2 Immune Diseases

The incidence of asthma, allergies and autoimmune diseases are increasing world-wide:

- 100–150 million people suffer from asthma world-wide, and the annual death rate has reached over 180,000 (WHO, 2002). In Western Europe, the incidence of asthma has doubled in the last 10 years (WHO, 2002). While most asthma is due to allergy, there are many cases where an allergic cause has not been established.
- 1 in 4 people in the UK (15 million) are affected by an allergy at some time in their lives. The incidence appears to be increasing at a rate of about 5% a year with as many as half of those now affected being children (Allergy UK, 2003).
- 1 in 31 people in the USA (over 8.5 million) have autoimmune diseases, which include diabetes, rheumatoid arthritis and multiple sclerosis, with the incidence in women occurring at 2.7 times the rate in men (Jacobsen et al., 1997).

While there is no doubt that genetic predisposition is an important determinant for immune disease, environmental factors do play a role in their development. Studies show that certain chemicals cause changes in the immune system – from changing immune parameters such as antibody levels and white blood cell numbers, which affects the body's ability to fight infections and cancer, to inducing allergy and autoimmune diseases.

A life-long capacity for immune response is determined during prenatal and early postnatal development (EHP, 1996; Luster et al., 2003). Impairment of the immune system can result from alterations in the development



Table 4.1
Infant mortality rates and percent of total infant deaths for the ten leading causes of infant deaths in 1999 in the USA.

All Causes  Congenital malformations, deformations	705.6	100
Congenital malformations deformations		.00
and chromosomal abnormalities	138.2	19.6
Disorders relating to short gestation and low birth weight, not classified elsewhere	110.9	15.7
Sudden infant death syndrome	66.9	9.5
Newborn affected by maternal complications of pregnancy	35.3	5.0
Respiratory distress syndrome	28.0	4.0
Newborn affected by complications of placenta, cord and membranes	25.9	3.7
Accidents (unintentional injuries)	21.3	3.0
Bacterial sepsis of the newborn	17.5	2.5
Diseases of the circulatory system	16.8	2.4
Atelectasis <sup>2</sup>	16.3	2.3
All other causes	228.3	32.4
	Disorders relating to short gestation and low birth weight, not classified elsewhere  Sudden infant death syndrome  Newborn affected by maternal complications of pregnancy  Respiratory distress syndrome  Newborn affected by complications of placenta, cord and membranes  Accidents (unintentional injuries)  Bacterial sepsis of the newborn  Diseases of the circulatory system  Atelectasis²	Disorders relating to short gestation and low birth weight, not classified elsewhere 110.9  Sudden infant death syndrome 66.9  Newborn affected by maternal complications of pregnancy 35.3  Respiratory distress syndrome 28.0  Newborn affected by complications of placenta, cord and membranes 25.9  Accidents (unintentional injuries) 21.3  Bacterial sepsis of the newborn 17.5  Diseases of the circulatory system 16.8  Atelectasis² 16.3

<sup>&</sup>lt;sup>1</sup>Deaths per 100,000 live births

(Data from Hoyert et al., 2001).

of the immune system and may be longlasting. The effects may not be manifested at birth and may not be expressed until adulthood (EHP, 1996).

Chemicals can also worsen the symptoms of immune diseases. For example, children are particularly susceptible to developing asthmatic symptoms and other respiratory problems when exposed to air pollution (Altshuler et al., 2003a).

#### 4.1.3 Cancers

Cancer is the cause of 26% of deaths in the UK and the second leading cause of death in the USA (Cancer Research UK, 2003b; Anderson, 2001). It is estimated that one in three people will develop cancer during their lifetimes. About 1.2 million people in the UK are currently diagnosed with cancer (Cancer Research UK, 2003a).

Cancer is the third most common cause of

death among children between 1–19 years (Anderson, 2001). The incidence of childhood cancer in the USA increased by 26% between 1975–1999. The biggest rise was estimated for brain and other nervous system cancers (50% increase) and acute lymphocyctic leukaemia (62% increase). Only 5–10% of cancers have been linked to genetic factors, the rest are influenced by a broad range of environmental factors.

Most childhood cancers differ from adult cancers and affect a wider range of cell types. Some are linked with specific genetic or chromosomal alterations that most likely occur at conception or shortly after (Altshuler et al., 2003a). Others appear to be related to the defective development, or dysgenesis, of organs or tissues (Sonnenschein & Soto, 1999). These tissues are considered to be in a pre-malignant state as they have a high risk of developing cancer. Some of these cancers, such as the liver tumour hepatoblastoma

<sup>&</sup>lt;sup>2</sup>Atelectasis is a collapse of lung tissue affecting part or all of one lung. Congenital atelectasis can result from a failure of the lungs to expand at birth and may be localised or may affect all of both lungs.



appear in earlier in childhood, while others, such as testicular cancer in undescended testes, may appear later. In contrast, adult cancers mostly affect the epithelial cells which line organs and structures, and are thought to result from the slow accumulation of genetic or structural damage due to ageing or to exposure to environmental factors. Sensitivity to cancer-causing effects of chemicals is much greater during the preand postnatal period.

#### 4.1.4 Nervous System Disorders

There appears to be an epidemic of developmental, learning and behavioural disabilities in children (Shettler et al., 2000). About 17% of children in the USA suffer from one of these disabilities. The number of children being treated for attention deficit disorder (ADD) and attention deficit hyperactivity disorder (ADHD) has increased dramatically in the last ten years (Houlihan et al., 2003; Shettler et al., 2000). The reported incidence of autism is also increasing (Houlihan et al., 2003). The causes are not known but chemical exposures are considered a potential contributor (Woodward, 2001).

Neurodegenerative diseases such as Parkinson's disease, may also be caused by chemicals. For example, exposure to heavy metals, pesticides, and tobacco smoke have been linked to Parkinson's disease (Siderowf & Stern, 2003; Woodward, 2001).

## 4.1.5 Developmental and Reproductive Disorders

Reproductive disorders, particularly in the male are increasing:

- Sperm counts in many parts of the world are decreasing by about 1% per year in industrialised countries. There are significant regional differences in sperm counts that cannot be explained by genetic factors (Swan et al., 2000).
- Incidence of hypospadias, a birth defect

- of the penis where opening of the urethra occurs at the bottom of the penis instead of the tip, doubled in the USA between 1970–1993 (Paulozzi et al., 1997).
- Cryptorchidism, where testicles fail to descend into the scrotum before birth, occurs in 2–5% of male babies in Western countries, and is increasing. Men born with this defect are also at a higher risk for testicular and breast cancer (Paulozzi, 1999).
- Testicular cancer is rising in some parts of the world. It is the most common cancer in men aged 20–34 years (Huyghe et al., 2003).

Testicular dysfunction, hypospadias, cryptorchidism and a higher risk of testis cancer are collectively known as 'testicular dysgenesis syndrome' and are recognised as likely to be caused by a disruption of sex hormones during development (Skakkebaek et al., 2001). The decreasing number of male births compared with females in many industrialised countries may also be linked to these defects in male reproduction (Davis et al., 1998; Grech et al., 2003).

There have also been decreases in the average age of puberty in some ethnic groups in the USA and other countries (Herman-Giddens et al., 1997; Krstevska-Konstantinova et al., 2001). An earlier age at puberty is associated with increased risk of impaired stature and early onset of risky behaviours, impairments in endocrine or reproductive function, and may be associated with an increased risk of cancer (Alshuler et al., 2003b).

A wide variety of chemicals have been identified that act as endocrine disrupters by mimicking or antagonising the effect of hormones, and/or disrupting the synthesis and metabolism of hormones and their receptors. Chemical endocrine disrupters may be playing a role in the rise of reproductive and developmental disorders.



## 4.2 Vulnerable Stages in Developing Cells

#### 4.2.1 Control of Cell Division

The process that a cell goes through to reproduce its DNA and divide to produce new cells is called the cell cycle. The cell undergoes four phases: gene expression and protein synthesis (G1) prepare the cell for the reproduction of its DNA (S), and then further protein synthesis and growth of the cell (G2) occurs prior to the final mitosis or dividing of the cell (M1). This process involves the interaction of many metabolic and control pathways, any of which could be targets for toxins. There are checkpoints throughout the cell cycle to prevent the cell moving into the next stage if previous stages are not completed properly. Each of the 210 cell types in the human body has its own cell cycle length, from a few hours to several months or more. A short cycle, with more rapid metabolic activity generally makes cells more vulnerable to toxic effects.

Embryonic cells generally have very short cell cycles. For example, in early pregnancy, the brain grows at a rate of 250,000 cells per minute (Chudler, 2003). A common regulatory failure in rapidly cycling embryonic cells is for G2 checkpoints to be bypassed. If checkpoint molecules have been inhibited by a toxin, the cell cycle may continue without all the specifications being met, which can lead to cell abnormalities and/or cell death (Altshuler et al., 2003b).

#### 4.2.2 Programmed Cell Death

Cell types and numbers in specific organs are controlled by both production and by removal through programmed cell death, or apoptosis. For example, apoptosis removes the webbing between the fingers, and removes populations of cells from the immune system that could cause autoimmune disease. Apoptosis plays a critical role in the developing nervous system, from early proliferation of brain cells and into postnatal life.

Disruption of normal patterns of apoptosis through altered gene expression or failure of signalling mechanisms is implicated in a wide range of diseases including autoimmune diseases and certain cancers. For example, the persistence of renal stem cells that should disappear 4-6 weeks before birth may make them vulnerable to postnatal exposures that transform them into Wilm's tumour, a relatively common childhood cancer (Sharpe & Franco, 1995). Disordered apoptosis during embryonic development may increase the risk of neurodegenerative illnesses such as Parkinson's and Alzheimer's diseases (Brill et al., 1999).

#### 4.2.3 Gene Expression

Gene expression, the translation of DNA into RNA and the transcription of RNA into proteins, controls cell division, apoptosis and metabolic activity of the cell. During development, gene expression is extraordinarily active. A high proportion of genes are being expressed and a large number of genes are being switched on or off to control the cell's activities. Such a high rate of activity provides a wide range of opportunities for chemicals to interfere with cell development by interacting directly with DNA to disturb gene expression, or with the expressed proteins such as enzymes or control molecules.

For instance, cell signalling within and between cells, is key to gene expression, cell migration, and other developmental mechanisms. Toxins that interfere with these vital molecular processes can cause permanent damage to a child's development. Brain development, for example, can be altered when the signal transduction of neurotransmitters is disrupted by the well-known toxins, alcohol and methylmercury (Altshuler et al., 2003b). Disruption of cell signalling processes has also been implicated in the development of cancer (Sonnenschein & Soto, 1999).



#### 4.2.4 Cell Differentiation and Maturation

Cells differentiate and mature to assume their specific form and function within their organ or tissue. This is under the control of inter- and intra-cell signalling. If cells fail to differentiate properly, following chemical or other interference, organ function may be compromised and foetal survival endangered. These undifferentiated cells are also more vulnerable to further effects of chemicals, particularly those that cause cancer. For example, developmental disturbances during liver organogenesis in the first trimester of pregnancy can lead to the embryonal liver tumour hepatoblastoma. Extremely low birth weight infants are 100 times more likely than normal birth weight infants to develop hepatoblastoma and its development has been linked to environmental causes (Ikeda et al., 1997; Maruyama, 1999).

## 4.3 Vulnerable Stages in Child Development

#### 4.3.1 Germ Cell Development

Sperm and egg cells (germ cells) begin developing in the foetus and mature during puberty. In the male the primordial germ cells develop *in utero*. From puberty they undergo constant cycles of growth, cell division, mitosis and meiosis to produce sperm. In females the germ cells undergo mitosis and the 1st phase of meiosis during foetal life. By puberty there are about 400,000 primary follicles. During each menstrual cycle, a group of follicles ripen, and one egg is released at ovulation.

Germ cells can be damaged during their development in the foetus, and during child-hood or adult life. Chemicals that harm germ cells can harm an adult's fertility and can result in congenital health problems in their children. The male reproductive system is particularly sensitive to chemicals due to their rapid cell cycle.

#### 4.3.2 Embryonic and Foetal Development

From the time the sperm and egg unite (conception) to birth, human life grows rapidly from a single cell zygote to an infant. Due to the complexity and speed of development and the high rate of growth throughout the prenatal period, this stage of development is more vulnerable to environmental exposures than any other period of development. There are three main stages of prenatal development:

- Periconceptual (weeks 1–2 following fertilisation)
- Embryonic weeks 3–7
- Foetal weeks 8–38.

During the periconceptual period, the zygote undergoes rapid cell division, implants into the wall of the uterus and forms the embryo. During this period, hazardous environmental exposures usually cause death rather than injury, resulting in spontaneous abortion (Moore & Persaud, 2003).

Most major organs begin to form during the embryonic period, with their growth and development continuing during the foetal stage and in infancy for some systems. The period of organ development varies from 3-8 weeks to 12-16 weeks depending on the organ system. The critical period for the development of the brain, for example, is the longest at 3-16 weeks. Disruption of development during this time can result in major disruptions in the largescale structure of organs of other structures. This may result in death but is more likely to form major physical malformations (congenital anomalies). Both the organ affected and the anomaly that results are highly dependent on both the chemical and the age of the embryo or foetus. For example, the drug diethylstilbestrol caused twice as many genetic abnormalities in male children if mothers took the medication before the eleventh week than after (Moore & Persaud, 2003).



During later stages of development, environmental exposures can result in impaired growth, physiological defects, or functional deficiencies. These effects may be manifest as low birth weight, prematurity, pregnancy complications, or late foetal death (Moore & Persaud, 2003).

#### 4.3.3 Infancy and Childhood

Major structures of the brain and other systems continue to develop throughout childhood. For example, in the brain and nervous system, neurone migration, cell proliferation, and synapse formation are all very active from birth through to 3 years. The development of cellular insulation around nerve fibres continues for at least 10 years (Rice & Barone, 2000).

The immune system develops extensively during childhood as immune memory (the ability to recognise and respond to foreign proteins and organisms) is established (Luster et al., 2003). Improper development of the immune system may` cause allergies and autoimmune diseases later in life.

#### 4.3.4 Puberty

Physical growth and maturation continue through puberty, particularly sexual maturation. This process is accompanied by complex interactions between the central nervous system and hormone-secreting organs, which can be affected by environmental factors.

#### 4.4 Health Effects of Contaminating Chemicals

A growing body of literature is now linking low-dose chemical studies in animals to a broad range of health effects that were previously unexplored in high dose studies. These are subtle but important changes in the functions of apparently undamaged organ systems including alterations in immune function, enzyme activity, hormone levels, neurobehavioral parameters, organ

growth, and neurotransmitter levels. Many low dose effects are detected following developmental exposure. Low doses can produce more serious effects during foetal development or infancy than similar exposures during adulthood. These studies are easier to relate to human health than high dose studies as they are within the dose range of normal hormone activities, and concentrations of contaminating chemicals in the body. Low dose studies also show that chemicals can produce a spectrum of health effects that vary with dose, and affect the target organ in different ways depending on the dose. For example, many hormone mimics produce opposite effects at low and high doses (known as a bisphasic response).

While low-dose animal studies are difficult to relate to humans, they are likely to be the best indication of toxicity we can have. Direct evidence of the contaminating chemicals on human health is difficult to find as there are no control groups to compare to we are all exposed to multiple chemicals and at widely varying levels. A second problem is that there is little data on the combined effects of chemicals - each chemical tends to be tested individually. A full understanding of their effects on humans would require testing all these compounds simultaneously at low levels. For example, one chemical might disturb the immune system while other chemicals cause cancerous changes to cells that then go unchecked by immune cells. Chemicals with similar effects, such as oestrogen mimics, together may be strong enough to unbalance hormone metabolism. The few studies that have combined two or more chemicals tend to support this presumption.

The POPs that follow are all toxic in animals. Evidence from low dose-studies varies considerably, but all show effects in animals, and occasional rare human studies, that are of serious concern to human health.



#### 4.4.1 Alkylphenols

These bioaccumulative chemicals are primarily hormone disrupters. Nonylphenol has recently been classified in the EU as a category 3 reproductive toxicant in terms of human fertility and development.

In sewage, alkylphenol ethoxylates degrade into alkylphenols. In the liver, the enzyme P-glycoprotein breaks down and protects organisms from the toxic alkylphenol ethoxylates, but the resulting alkylphenols are oestrogenic (Loo and Clarke, 1998). Octyl- and nonylphenols are the most studied of this group of chemicals.

Both nonylphenol and octylphenol show oestrogenic and anti-androgenic activities (Lee et al., 2003a; Paris et al., 2002). Despite having a lower binding affinity to oestrogen receptors, nonylphenol is a more powerful oestrogen than octylphenol as serum has more of a protective effect against octylphenol. This is an important finding to consider when comparing dietary levels and actual effects of these and other oestrogen mimics (Nagel et al., 1997 and 1999).

#### **Reproductive and Developmental Toxins**

Two low-dose developmental studies in rodents provide examples of the effects of alkylphenols on reproduction and development. Sharpe et al. (1995) showed that preand postnatal exposure to octylphenols caused a reproducible and consistent decrease in testicular size and daily sperm production in rats during a relatively short period (Table 4.2). A multigenerational mouse study demonstrated that nonylphenol affected both the parents and offspring (Kyselova et al., 2003) with the predominant effects being on the size of male reproductive organs, sperm quality and fertility (Table 4.3).

#### **Immunotoxins**

Preliminary studies show that nonylphenol may also disrupt the human immune system. For example, in vitro studies in human breast cells show that nonylphenol inhibits the production of a monocyte chemokine, a chemical which attracts and activates monocytes, an important group of white blood cells (Inadera et al., 2000). In mice, nonylphenol also increases the production of the chemical messenger interleukin-4 (IL-4) in T lymphocytes, and increases levels of IgE antibodies. Both IL-4 and IgE are very important in

Table 4.2

Reproductive effects of 4-octylphenol following prenatal and postnatal exposure via drinking water.

Exposure	Levels (µg/L) <sup>1</sup>	Effects <sup>2</sup>
Postnatal	100	increased body weight
		increased kidney weight
		decreased testes weight
Foetal & postnatal	100	decreased kidney weight
		decreased testes weight
		decreased prostate weight
Postnatal	1000	decreased testes weight
		increased kidney weight
Foetal & postnatal	1000	decreased testes weight
		decreased prostate weight
		decreased sperm production

<sup>&</sup>lt;sup>1</sup>Mothers' intake levels of 125mg/kg/day for up to 2 days after birth to 370 mg/kg/day just before weaning for the 1000mg/L group are likely to be overestimates of intake.

(Data from Sharpe et al., 1995)

<sup>&</sup>lt;sup>2</sup>Changes to organ weights are all relative to body weight.



Table 4.3

Reproductive effects of nonylphenol following multigenerational exposure via drinking water.

Generation	Levels (µg/L) <sup>1</sup>	Effects <sup>2</sup>
Parent male	50	decreased testes weight decreased kidney weight
		14% sperm damage
	500	increased epididymis weight
		10% sperm damage
Parent female	50	decreased body weight
		decreased kidney weight
	500	decreased kidney weight
F1 male offspring	50	decreased prostate weight
		26% decreased liver weight
	500	increased body weight
		decreased testes weight
		26% decreased liver weight
F2 male offspring	50	decreased litter size
, 0	500	decreased litter size

<sup>&</sup>lt;sup>1</sup>Actual intake levels not reported.

(Data from Kyselova et al., 2003)

allergy, and nonylphenol may therefore enhance allergic responses (Lee et al., 2003b).

#### 4.4.2 Bisphenol A

Bisphenol A is another EU category 3 reproductive toxicant. Bisphenol A binds to oestrogen receptors range of human cell lines and mimics all the oestrogenicity parameters, confirming it as one of the stronger oestrogenic chemicals (Meerts et al., 2001; Olsen, 2003). In fact, although bisphenol A shows a lower binding affinity than octylphenol *in vitro*, its oestrogenic activity in mice is predicted to be 500 times more powerful due to interactions with serum (Nagel et al., 1997 and 1999).

Bisphenol A follows the classic U-shaped dose curve seen for a range of hormones and hormone mimics. For example, in embryonic cells, low concentrations of bisphenol A increase the development rate while concentrations 100,000 times higher decrease developmental rate (Takai et al., 2000 and 2001). In prostate cancer cells, bisphenol A increases cell proliferation at concentrations 100 times lower than the levels that inhibit cell growth (Wetherill et al., 2002).

#### **Developmental and Reproductive Toxin**

The health effects of bisphenol A have been demonstrated in an ever-increasing number of animals studies at levels 2,500 times lower than the EPA's 'lowest observed dose effect' dose. The adverse outcomes range from altered male reproductive organs and aggressive behaviour, to abnormal mammary gland growth, early puberty and reduced breast feeding (Table 4.4). Human infants ingest bisphenol A in formula at an estimated daily rate of 1.6 µg/kg/day, giving little safety margin for doses that cause effects in animals (as little as 2 µg/kg/day) (Houlihan et al., 2003).

#### Immunotoxin

Various studies show that bisphenol A can also affect the immune system by:

- reducing production of mouse macrophage nitric oxide and tumour necrosis factor-, a chemical that inhibits tumour growth (Kim & Jeong, 2003).
- inhibiting the production of a monocyte chemokine by human breast cancer cells, a chemical that attracts and activates monocytes (Inadera et al., 2000).

<sup>&</sup>lt;sup>2</sup>Changes to organ weights are all relative to body weight.



Table 4.4
Assumed safe dose of bisphenol A compared to evidence of low-dose toxicity in rodents.

Effects	Dose (mg/kg/day)	Reference
Assumed safe dose for animals	5.0	US EPA, 1993
Assumed safe dose for humans*	0.05	US EPA, 1993
Effects on vagina	0.100	Schonfelder et al., 2002a
Increased prostate size	0.050	Gupta et al., 2000
Long-term alterations in behavioural		
patterns in adolescence and adulthood	0.040	Adriani et al., 2003
Abnormal prostate development	0.025	Ramos et al., 2001
		Ramos et al., 2003
Abnormal mammary gland growth		
(changes associated with carcinogenesis).	0.025	Markey et al., 2001
Reduced sperm production	0.020	vom Saal et al.,
		1998 Sakaue et al., 2001
Early puberty in females	0.020	Honma et al., 2002
Altered maternal care	0.010	Palanza et al., 2002
Early puberty in females	0.0024	Howdeshell et al., 1999
Altered male reproductive glands	0.002	vom Saal et al., 1998
Increased adult prostate weight	0.002	Nagel et al., 1997
· · · · · ·		Nagel et al., 1999
Reduced testis weight	0.002	Kawai et al., 2003
Decreased antioxidant enzymes in the liver	0.002	Bindhumol et al., 2003
Activated temporary aggressive behaviour	0.002	Kawai et al., 2003

<sup>\*</sup>The oral Reference Dose (RfD) is an estimate of a daily exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious effects during a lifetime.

- enhancing proliferative responses and cytokine production of T lymphocytes (Yamashita et al., 2003)
- increasing IL-4 production in mouse T cells and increasing levels of IgE antibodies levels, both important factors in allergy (Lee et al., 2003b).

#### **Human Studies**

The few studies available in humans do link bisphenol A to human health. Serum bisphenol A levels are higher in men as well as in women with polycystic ovary syndrome compared to normal women. These levels correlated with testosterone levels indicating that there may be differences in androgen-related metabolism of bisphenol A (Takeuchi & Tsutsumi, 2002). This is a factor to consider when comparing

exposure levels to possible health effects.

Bisphenol A has also been found in ovarian follicular fluid of women requiring IVF treatment (Ikezuki et al., 2002). Mothers carrying foetuses with abnormal chromosomes had higher serum bisphenol A concentrations than those with normal foetuses (Yamada et al., 2002).

Concentrations of bisphenol A have been found at levels five times higher in amniotic fluid at 15–18 weeks gestation than in other fluids (Ikezuki et al., 2002). This is a vital period of organ development and highlights the increased exposure and susceptibility of the developing foetus to chemicals that contaminate pregnant women.

#### GREENPEACE

#### 4.4.3 Brominated Flame Retardants

Most brominated flame retardants (BFRs) are persistent and/or bioaccumulative and several are endocrine disrupters. They are widespread as far as the Arctic where their concentrations in seals are increasing exponentially (Ikonomou et al., 2002).

Toxicity data on BFRs are severely limited and focus primarily on high dose studies of PBDEs in animals. The toxic profile of the PBDEs is proving to be very similar to PCBs. Their toxicity includes birth defects, liver and kidney damage, thyroid imbalances and neurological damage to animals and humans. Generally, the penta-BDEs seem to be the most toxic (Darnerud et al., 2001; Darnerud, 2003).

Data on TBBP-A and HBCD are almost completely lacking. TBBP-A *in vitro* studies indicate toxic effects on the immune system and thyroid, while HBCD seems to affect the liver and thyroid (Darnerud et al., 2001; Darnerud, 2003).

#### **Thyroid Toxins**

The similarity of PBDEs to PCBs and the thyroid hormone may underlie their toxicity: PCBs, PBDEs and thyroid hormones all consist of two six-carbon rings decorated with halogens of chlorine, bromine or iodine respectively (Fig. 4.1). In vitro studies show that PBDE metabolites, as well as pentabromophenol (PBP) and TBBP-A, compete strongly with the thyroid hormone, thyroxin, for binding to the thyroxin serum binding protein, transthyretin (Meerts et al., 2000). It is likely that in the body, PBDEs and their metabolites displace thyroid hormones from transthyretin thereby allowing increased metabolism of the hormone and thus decreased levels in serum. The authors speculate that the effects of BFRs on the thyroid in vivo may be comparable to the thyroiddisrupting effects of PCBs (Meerts et al., 2000). As the development of the central nervous system is highly dependent on thyroid hormones, alterations in thyroid

Figure 4.1

The structural similarity of the brominated flame retardants, polybrominated diphenyl ethers (PBDEs), to the polychlorinated bisphenols (PCBs) and thyroid hormones may underlie their toxicity.

2,2',4,4',5,5'-hexachlorobiphenyl (PCB-153)

### 2,2',4,4',6-pentabromo diphenylether (BDE-100)

#### Thyroxine (T4)

homeostasis may result in permanent neurobehavioral defects.

#### Neurotoxins

PBDEs, such as BDE-209, can be taken up during neonatal life, distributed throughout the body and concentrate in the brain. They induce developmental neurotoxic effects in adult mice that worsen with age and lead to abnormal behaviour. The effects seem to be inducible only during a defined critical period of neonatal life (Viberg et al., 2003). For example, perinatal exposure to PBDEs at levels from 0.6 mg/kg produced several behavioural alterations in mice with the main



effect being marked hyperactivity at adult-hood (Branchi et al., 2002 and 2003). Learning and memory deficits have also been found at adulthood for neonatally exposed animals (Eriksson et al., 2001). A recent study shows that HBCD also has behavioural effects on neonatal mice at 0.9 mg/kg (cited in Darnerud, 2003).

A rat study showed that HBCD and TBBP-A inhibit the uptake of the neurotransmitters dopamine, glutamate and gamma-amino-n-butyric acid (GABA) by neurones at a concentration level similar to those found with studies on PCBs and ecstasy (Mariussen & Fonnum, 2003).

#### **Oestrogenic Effects**

A range of PBDEs and brominated bisphenol A compounds, such as TBBP-A, show oestrogenicity in human cells lines and bind to the oestrogen receptors (Meerts et al., 2001). The metabolism of PBDEs to hydroxylated-PBDEs produce more potent oestrogen mimics. Brominated bisphenol A compounds with the lowest bromination showed the highest effect, and among the PBDEs, BDE-100, BDE-75 and BDE-51 showed the highest activity. TBBP-A does not mimic all the oestrogen test parameters and, therefore, could cause an unbalanced oestrogen response (Olsen et al., 2003).

#### **Cancer Promoters**

Three PBDEs (mono-, di-, and tetra-BDE) induce intragenic genetic recombination in mammalian cells, which is known to provoke a number of diseases, including cancer (Helleday et al., 1999). Tris(2,3-dibromo-propyl)phosphate and its metabolite 2-bromoacrolein may be clastogens – chemicals which cause breakage in chromosomes (van Beerendonk et al., 1994).

#### 4.4.4 Organotins

Organotins are persistent chemicals that accumulate throughout the body, primarily in the liver and kidneys. The tri-tins, such as TBT

and TPT, are all listed as poisons and described as respiratory toxins, fetotoxins, reproductive toxins, immunotoxins, possible carcinogens, skin and respiratory irritants, and allergens (Norris, 1994; van Heijst, 1994).

The toxicity of organotins stems primarily from the fact that they are powerful metabolic inhibitors. They inhibit a range of enzymes responsible for the metabolism, signalling, and regulation of many cell activities. This can cause a variety of effects at various concentrations depending on the target organ. (Norris, 1994).

#### **Reproductive and Developmental Toxins**

Organotins are well-known hormone disrupters with devastating effects on marine molluscs. TBT and TPT inhibit a variety of enzymes responsible for the production of male and female sex steroid hormones – oestogens, testosterone, and oestradiol (Doering et al., 2002; Lo et al., 2003; Steckelbroek et al., 2001). While low-dose developmental studies are lacking in mammals, insufficient activation of male hormones is known to be responsible for developmental disorders of the male reproductive tract.

A recent study suggests that organotins may also cause developmental effects *in utero* at relatively low doses by targeting the maternal thyroid (Adeeko et al., 2003). Effects varied depending on dosage but appeared to be linked to the reduction of maternal serum thyroxine and triiodothyronine throughout gestation. Effects included reduced maternal weight gain; increased post-implantation loss; decreased litter sizes; decreased foetal weights; delayed foetal skeletal development; and abnormalities in genital development in male fetuses.

#### **Immunotoxins**

Organotins are well known immunotoxins which target a range of white blood cells and destroy or limit their ability to function. For example:



- DBT and TBT both suppress mitogenesis in human lymphocytes. This process of stimulating lymphocytes to undergo mitosis is vital for the reproduction of lymphocytes during the immune response to invasion by organisms such as bacteria or viruses (Nakata et al., 2002).
- TBT causes apoptosis (or self-destruction) of human T lymphocytes (Stridh et al., 2001).
- BPT, TBT and TPT all inhibit the cytotoxic function of human NK cells, which are responsible for killing tumour cells and cells that are infected by viruses (Whalen & Loganathan, 2001; Whalen et al., 2002a and b; Whalen et al., 2003).
- TPT enhances neutrophil maturation (Watanabe et al., 2003). It also inhibits superoxide production by human neutrophils, which thereby limits the ability of neutrophils to kill bacteria (Miura & Matsui, 1991).

### **Cancer Promoters**

TPT is a potent spindle poison – a toxin that interferes with the cell components responsible for moving chromosomes into place before a cell divides. TPT acts synergistically with pentachlorobiphenyl, a PCB, to induce abnormal chromosome arrangements in mitosis at very low concentrations (10nM and 50nM) (Jensen et al., 2000).

## 4.4.5 Phthalates

Toxicity studies primarily focus on DEHP, the most toxic phthalate, and DINP, and the application of these profiles to humans remain controversial (CDC, 2003; Lovekamp-Swan & Davis, 2003; Shea, 2003).

Phthalates are hormone disrupters that act as anti-androgens. The diester forms of phthalates are converted to monoester forms in the gut, liver and blood. These are considered the ultimate toxins, and the length and structure of the side chain is important for their levels of toxicity (Lovekamp-Swan & Davis, 2003).

The primary mechanism of toxicity of phthalates appears to be via the activation of a set of proteins called PPARs - nuclear transcription proteins that regulate a wide variety of cell activities (Boiter et al., 2003; Lovekamp-Swan & Davis, 2003). PPAR is found in tissues that are highly metabolically active, such as the liver. Activation of PPAR by DEHP/MEHP results in decreased transcription of aromatase, the enzyme which converts testosterone to estradiol, and therefore disrupts sex hormone metabolism. PPAR<sub>v</sub> is highly expressed in various human tissues including adipose tissue and immune cells. PPAR<sub>v</sub> is a key regulator of cell differentiation. By activating PPAR<sub>v</sub>, DEHP/ MEHP disrupts, for example, the critical growth and differentiation of the ovarian follicle.

Phthalates also have toxic effects that are at least partly independent of PPAR activation. Testicular toxicity is partly caused by interference with binding of follicle stimulating hormone to its receptor on Sertolli cells (Boiter et al., 2003). Phthalates may also bind to the oestrogen receptor. For example, the effects of 19 phthalates in human breast cancer cells (Okubo et al., 2003) showed that:

- DCHP, DEHP and BBP are oestrogenic
- DCHP, MOP and MEHP are cytotoxic
- MMP, MCHP, MBZP and MIPrP are antioestrogenic.

## **Reproductive and Developmental Toxins**

Phthalates, following their conversion to a toxic metabolite, can produce foetal death, malformations, and reproductive toxicity, with different profiles for each phthalate and different potencies (Shea, 2003).

Maternal exposure of rodents to DEHP/MEHP, causes reduced embryonic implantation; increased resorptions; skeletal, cardiovascular and eye abnormalities; neural tube defects; intrauterine death; increased postnatal death; and decreased intrauterine and postnatal growth in rodent



pups (Gray, 2000; Moore et al., 2001; Shea, 2003). Foetal toxicity can occur without evidence of maternal toxicity. The immature male reproductive tract is the most sensitive system. Pathological changes in the testis, and decreased sperm numbers are consistent effects. Prenatal and postnatal exposure leads to complete female infertility and decreased male fertility. Sertoli cells in the testicle and the granulosa cells of pre-ovulatory follicles in the ovary appear to be the primary cellular target of DEHP/MEHP.

Other phthalates appear to have a similar pattern of toxicity but at higher doses. For example:

- DINP causes skeletal and genito-urinary abnormalities (Shea, 2003).
- DBP is a testicular toxicant and causes reproductive tract malformations in male rats after *in utero* exposure (Lovekamp-Swan & Davis, 2003).
- Maternal exposure to low doses of BBP (125–370 μg/kg/day) lead to decreased testes weight in male offspring following prenatal and postnatal exposure (Sharpe et al., 1995).
- DPP and DHP cause testicular atrophy and are both female and male reproductive toxicants (Lovekamp-Swan & Davis, 2003; Shea, 2003).

#### **Cancer Promoters**

The carcinogenic effects of DEHP in liver are thought to be due to PPAR activation (Lovekamp-Swan & Davis, 2003). Humans have one-tenth the level of PPAR expression in liver compared to mice so the carcinogenicity in humans is under dispute. The activation of enzymes by PPAR may also increase susceptibility to other chemicals whose metabolites are toxic, including carcinogens.

PPAR<sub>y</sub> plays a critical role in adipocyte differentiation, therefore DEHP activation of PPAR<sub>y</sub> in tissues other than liver could alter normal pathways of differentiation. This could explain the teratogenic effects of DEHP as development is a critical time for differentiation, and suggests that it could also play a role in cancers (Lovekamp-Swan & Davis, 2003).

### **Immunotoxins**

There is some evidence of immunotoxicity which may be related to PPAR<sub>y</sub> on immune cells. PPAR<sub>y</sub> activation profoundly alters the immune properties of these cells, usually leading to the inhibition of immune responses (Nencioni et al., 2003). For example, phthalates can disrupt normal antibody production. In one study, various phthalate metabolites either stimulated or suppressed IgG and IgE antibody produc-

Table 4.5

Different phthalate metabolites can both stimulate and suppress antibody production depending on their concentration levels (μg/mL).

Phthalate Metabolite	Decreased Antibody Production		Increased Antibody Production		
	IgE	lgG1	IgE	lgG1	
MBnP	ns	ns	ns	ns	
MEHP	1000	1000	10	ns	
MiDP	100	100	ns	ns	
MiNP	1000	10	100	ns	
MnBP	ns	ns	ns	ns	
MnOP	1000	1000	ns	10 and 100	

IgG1: Immunoglobin G1; IgE: Immunoglobin E; MBnP: monobenzylphthalate; MEHP: mono-2-ethylhexyl phthalate; MiDP: mono-iso-decylphthalate; MiNP: mono-iso-nonylphthalate; MnBP: mono-n-butylphthalate; MnOP: mono-n-octylphthalate; NS: no significant effect.

(Data from Larsen et al., 2001)



tion depending on their concentration levels (Larsen et al., 2001). Interestingly, MEHP induced IgE production at relatively low levels – the antibody responsible for allergic reactions (Table 4.5).

#### **Human Studies**

An interesting study in Puerto Rico demonstrated a link between the increasing incidence of premature breast development (thelarche) and exposure to phthalates (Colon et al., 2000). The annual incidence of thelarche in Puerto Rican girls aged 6-24 months was 8 cases per 1000 female births from 1984-1993, the highest ever reported. High levels of phthalates were detected in 68% of all thelarche girls but only 17% of normal girls. The authors suggest that the high incidence of thelarche in Puerto Rico may be caused by a high exposure to phthalates due to high importation of plastic packaged foods, with a year-round high temperature and humidity which promotes closed environments and frequent use of air-conditioners.

Women with endometriosis have a higher level of DEHP in their blood than normal women, and 92.6% of these also had detectable DEHP and/or the metabolite MEHP in their peritoneal fluid (see Table 2.5, Section 2). This suggests that DEHP plays a role in the pathogenesis of endometriosis (Corbellis et al., 2003).

Chewing on plastics that contain phthalates may increase the risk of developing tumours in the mouth, and upper airway and digestive tracts, according to one study, which showed that DBP and DiBP cause significant DNA damage to human tonsils *in vitro* (Kleinsasser et al., 2001).

Patients exposed to DEHP as the consequence of medial treatment with devices containing PVC may provide evidence of the human toxicity of phthalates. Virtually all medical devices made from PVC contain

DEHP. Patients on dialysis, blood transfusions, artificial ventilation, and exchange transfusions are at risk. Premature infants are particularly at risk as they are exposed to relatively high amounts of DEHP while in intensive care. Reports of possible links between DEHP and disease in humans include (EC, 2002):

- development of polycystic kidney disease in haemodialysis patients
- pulmonary toxicity in mechanically ventilated pre-term infants with PVC tubes
- cholestasis (bile flow stops) in infants with serum levels of 18–98 μg/mL DEHP following the use of an artificial lung (extracorporeal membrane oxygenation therapy)
- abnormalities in livers in patients after a year of haemodialysis
- liver toxicity in Rhesus monkeys receiving blood plasma from PVC bags over a year.

As described in Section 4.2.4, extremely low birth weight (ELBW) infants are 100 times more likely to develop the liver tumour hepatoblastoma. While ELBW infants may develop this tumour anyway, exposure to high levels of DEHP while in intensive care may also play a role in its development (EC, 2002).

# 4.4.6 Artificial Musks

Artificial musks are persistent and bioaccumulative chemicals. Most health studies focus on musk xylene (MX) and musk ketone (MK).

Although MX and MK are structurally similar and possess almost identical physico-chemical properties, they differ significantly in their biological properties. Both musks possess oestrogenic activity *in vitro* with MK showing an affinity for the oestrogen receptor three times greater than MX (Bitsch et al., 2002). However, when MK is reduced to its metabolite it loses its activity, whereas when MX is converted to p-amino-musk xylene, its oestrogenic potency increases (Bitsch et al., 2002).



MX appears to have a higher half-life in humans than rodents, and induces the liver CYP 2B enzymes in rat and mice (which its metabolite then inhibits) (Schmeiser et al., 2001). MX has been detected chemically bound to haemoglobin (Hb-adducts) in the blood of ten volunteers (Riedel et al., 1999). The ability to form Hb-adducts correlates well with the formation of DNA adducts which cause tumours, suggesting a cancer risk for substances causing Hb-adducts. Chronic exposure to MX is also known to cause mouse liver tumours (Schmeiser et al., 2001).

Few toxicological studies on MK are available. However, it shows a strong induction of liver enzyme activities in rats which are distinct from those inhibited by MX (CYP 1A1 and 2, and CYP 2B). These liver enzymes metabolise a range of chemicals and drugs, turning some of them, such as polycyclic aromatic hydrocarbons, into carcinogens and mutagens. Exposure of humans to MK might increase their susceptibility to the hazards of other cancer-causing chemicals. For example, MK acts as a co-genotoxicant by amplifying the DNA damaging effects of the polycyclic aromatic hydrocarbon, benzo(a)pyrene, in human cells at relatively low doses (Mersch-Sundermann et al., 2001).

The polycyclic musks, AHTN and HHCB, are selective oestrogen receptor modulators, and induce both oestrogenic and anti-oestrogenic activity depending on the cell type and the receptor subtype targeted. Weak oestrogenic effects are observed at relatively high concentrations (10  $\mu$ M) while anti-oestrogenic effects are seen at 0.1  $\mu$ M (Schreurs et al., 2002).

One human study has shown a significant association between MX and MK levels in blood and hormonal and gynaecological

problems in women suggesting that these musks cause reproductive toxicity and endocrine effects in humans (Eisenhardt et al., 2001).

## 4.4.7 Chlorinated Paraffins

Chlorinated paraffins (CPs) are persistent and bioaccumulative; however, there is very little data available on low-dose chronic effects in animals, and none available on their effects in humans. The data available suggests that theses chemicals may act as carcinogens and disrupt thyroid hormones in humans.

The toxicity of two CPs are fairly well characterised: C12 60% and C23 43%. The C12 60% appears to have a greater potential for chronic toxicity and carcinogenicity than C23 43%. The C12 60% is toxic to liver, lymph glands, kidney, and thyroid; causes benign tumours in kidney, liver, and thyroid; and causes malignancies in kidney, and leukaemia. C23 43% is associated with malignant lymphomas (Bucher et al., 1987; NTP, 1986a; NTP, 1986b).

Short and intermediate, but not long carbon chain CPs, are potent inhibitors of intercellular communication, which may be their mechanism for promoting tumours (Kato & Kenne, 1996; Warngard et al., 1996).

One study has shown that low levels (50 mg/kg) of medium length CPs (C14–17, 52%) are toxic to liver and thyroid in female rats (Poon et al., 1995). A rare study looking at the affects of two chemicals (at mg/kg levels) on thyroid health demonstrated a synergistic effect between the flame retardant BDE-47 and the CP known commercially as 'Witaclor 171P' in decreasing free plasma thyroxine in rats (Hallgren & Darnerud, 2002).



# 5 CONCLUSION

While this report was not intended to be a comprehensive literature review, it does provide ample evidence that:

- Many POPs are now being discovered in human tissues, including breast milk, and the blood of foetuses and newborn children (for a summary see Table 5.1). Given the wide range of everyday products containing these POPs and their contamination of the environment, it is likely that further evidence of contamination is forthcoming.
- Unborn and newborn children are particularly at risk of exposure to POPs as they absorb chemicals more efficiently, process them more slowly and eliminate them less efficiently.
- There are a number of non-infectious diseases on the rise, particularly in industrial countries. These include congenital defects, immune diseases, developmental and reproductive disorders, neurological disorders and cancers. These diseases often begin in childhood, and may be caused by damage to the developing child the period of life most susceptible to chemical damage.
- While the causes of these diseases are not clear, there is a general concern within the scientific and medical communities that chemicals are contributing to an increase in these diseases.
- The laboratory evidence of toxicity varies widely for each group of POPs, but there

- is increasing evidence that low doses of these chemicals are linked with a broad range of health effects that were previously unexplored in high-dose studies.
- Direct evidence of the effects of POPs on human health is lacking.

It is clear that the 'evidence' generally regarded as proof of harm to human health is not available for POPs, and even with further research is unlikely to be. Strong links between POPs and disease are very difficult to establish: there are no uncontaminated control groups for comparison; many of these diseases do not become apparent until long after chemical exposure begins; actual exposure levels are very difficult to estimate and are likely to vary considerably from person to person and over an individual's lifetime; and very little is known about the effects of exposure to chemical mixtures.

Waiting for further 'solid' evidence of chemical effects on health will mean risking irreversible damage to the health of further generations of children.

Continued research into the health effects of POPs is certainly required in order to better understand the causes of non-infectious diseases and, hopefully, to monitor their decline as these POPs are removed from products or substituted with safer alternatives. This may be the best evidence of their effects on health we can provide.



Table 5.1

Summary of the possible health effects of the chemical contamination of the child.

Chemical group and examples	Found in	Laboratory evidence	Human evidence	Likely effects on child health
Alkylphenols Octylphenol Nonylphenol	Umbilical cords Breast milk	Oestrogen mimics Immunotoxins		Developmental & repro ductive disorders Immune disorders
Bisphenol A	Umbilical cords Umbilical cord blood Amniotic fluid Placental tissue Breast milk Adult ovaries Adult blood	Oestrogen mimic Immunotoxin	Linked to polycystic ovary syndrome, female fertility problems, & abnormal foetal chromosomes.	Developmental & repro ductive disorders Immune disorders
Brominated flame retardants PBDEs TBBP-A HBCD	Umbilical cord blood Breast milk Breast fat Adult blood Adult fat	Thyroid hormone disrupters Oestrogen mimics Neurotoxins Cancer promoters		Developmental & repro ductive disorders Nervous system disorders Cancers
Organotins Dibutyltin Tributyltin Triphenyltin	Adult blood Adult liver	Enzyme inhibitors Hormone disrupters Immunotoxins Cancer promoters		Developmental & repro ductive disorders Immune disorders Cancers
Phthalates DEHP DINP	Child blood & urine Adult blood & urine	Enzyme disrupters Hormone disrupters Immunotoxins Cancer promoters	Linked to premature breast development & endometriosis. DEHP in medical devices linked to liver, kidney & respiratory diseases.	Developmental & repro ductive disorders Immune disorders Cancers
Artificial musks Musk xylene Musk ketone AHTN HHCB	Breast milk Adult blood Adult fat	Enzyme inducers Hormone disrupters	Linked to hormonal & gynaecological problems in women	Developmental & repro ductive disorders Cancers
Chlorinated paraffins C12 60% C23 43%	Adult fat	Inhibit intercellular communication Toxic to liver, kidney, thyroid, & lymph glands Cancer promoters		Cancers



# 6 REFERENCES

Adeeko A, Li D, Forsyth DS, Casey V, Cooke GM, Barthelemy J, Cyr DG, Trasler JM, Robaire B, Hales BF (2003). Effects of *in utero* tributyltin chloride exposure in the rat on pregnancy outcome. *Toxicol Sci*;74(2):407–15.

Adriani W, Seta DD, Dessi-Fulgheri F, Farabollini F, Laviola G (2003). Altered profiles of spontaneous novelty seeking, impulsive behavior, and response to D-amphetamine in rats perinatally exposed to bisphenol A. *Environ Health Perspect*;111(4):395–401.

Allergy UK (2003). Stolen Lives – The Allergy Report. *Allergy UK*; May 2003. Accessed Sept 2003 at: http://www.allergyfoundation.com/stolenlives.html

Altshuler K, Berg M, Frazier LM, Laurenson J, Mendez W, Molgaard CA (2003a). Overview of the special vulnerability and health problems of children. OCHP Paper Series on Children's Health and the Environment. Washington, DC, USA: US Environmental Protection Agency. Paper 2003-1. Accessed Sept 2003 at: http://yosemite.epa.gov/ochp/ochpweb.nsf/content/1\_Intro.htm

Altshuler K, Berg M, Frazier LM, Laurenson J, Longstreth J, Mendez W, Molgaard CA. (2003b). Critical periods in development. OCHP Paper Series on Children's Health and the Environment. Washington, DC, USA: US Environmental Protection Agency. Paper 2003-2. Accessed Sept 2003 at: http://yosemite.epa.gov/ochp/ochpweb.nsf/content/2\_Intro.htm

Altshuler K, Berg M, Frazier LM, Laurenson J, Mendez W, Molgaard CA (2003c). Children's environmental exposures. *OCHP Paper Series on Children's Health and the Environment*.

Washington, DC, USA: US
Environmental Protection Agency.
Paper 2003-3. Accessed Sept 2003 at: http://yosemite.epa.gov/ochp/ochpweb.nsf/content/3\_Intro.htm

Anderson RN (2001). Death: Leading causes for 1999. *National Vital Statistics Report*; 49 (11);1–88. Hyattsville, MA, USA: National Centre for Health Statistics. Accessed Sept 2003 at: http://www.cdc.gov/nchs/ data/nvsr/nvsr49/nvsr49\_11.pdf

Angerer J, Kafferlein HU (1997). Gas chromatographic method using electron-capture detection for the determination of musk xylene in human blood samples. Biological monitoring of the general population. *J Chromatogr B Biomed Sci Appl*;693(1):71–8.

Barr DB, Silva MJ, Kato K, Reidy JA, Malek NA, Hurtz D, Sadowski M, Needham LL, Calafat AM (2003). Assessing human exposure to phthalates using monoesters and their oxidized metabolites as biomarkers. *Environ Health*Perspect;111(9):1148–51.

Bearer CF (1995). How are children different from adults? *Environ Health Perspect*; 103 (Suppl 6):7–12.

Bindhumol V, Chitra KC, Mathur PP (2003). Bisphenol A induces reactive oxygen species generation

in the liver of male rats. *Toxicology*;188(2–3):117–24.

Bitsch N, Dudas C, Korner W, Failing K, Biselli S, Rimkus G, Brunn H (2002). Estrogenic activity of musk fragrances detected by the E-screen assay using human mcf-7 cells. *Arch Environ Contam Toxicol*; 43 (3):257–64.

Boitier E, Gautier JC, Roberts R (2003). Advances in understanding the regulation of apoptosis and mitosis by peroxisome-proliferator activated receptors in preclinical models: Relevance for human health and disease. *Comp Hepatol*;2(1):3.

Bouma K, Schakel DJ (2002). Migration of phthalates from PVC toys into saliva simulant by dynamic extraction. *Food Addit Contam*;19(6):602–10.

Branchi I, Alleva E, Costa LG (2002). Effects of perinatal exposure to a polybrominated diphenyl ether (PBDE 99) on mouse neurobehavioural development. *Neurotoxicology*; 23 (3):375–84.

Branchi I, Capone F, Alleva E, Costa LG (2003). Polybrominated diphenyl ethers: Neurobehavioral effects following developmental exposure. *Neurotoxicology*; 24 (3):449–62.

Brill A, Torchinsky A, Carp H, Toder V (1999). The role of apoptosis in normal and abnormal embryonic development. *J Assist Reprod Genetics*;16:512–9.



Brede C, Fjeldal P, Skjevrak I, Herikstad H (2003). Increased migration levels of bisphenol A from polycarbonate baby bottles after dishwashing, boiling and brushing. *Food Addit Contam*;20(7):684–9.

Bucher JR, Alison RH, Montgomery CA, Huff J, Haseman JK, Farnell D, Thompson R, Prejean JD (1987). Comparative toxicity and carcinogenicity of two chlorinated paraffins in F344/N rats and B6C3F1 mice. Fundam Appl Toxicol;9(3):454–68.

Cancer Research UK (2003a).

Cancer Stats. Incidence – UK.

London, UK: Cancer Research

UK. Accessed Sept 2003 at:

http://www.cancerresearchuk.org/
aboutcancer/statistics/
cancerstatsreport/

Cancer Research UK (2003b).

Cancer Stats. Mortality – UK.

London, UK: Cancer Research

UK. Accessed Sept 2003 at:

http://www.cancerresearchuk.org/
aboutcancer/statistics/

cancerstatsreport/

CDC (2003). Second National Report on Human Exposure to Environmental Chemicals. (Revised version). Atlanta, GA, USA: Centres for Disease Control and Prevention, National Center for Environmental Health. NCEH Pub. No. 02-0716. Accessed Sept 2003 at: http://www.cdc.gov/exposurereport/pdf/secondNER.pdf

Champ MA (2000). A review of organotin regulatory strategies, pending actions, related costs and

benefits. *Sci Total* Environ;258(1-2):21-71.

Choi JW, Fujimaki TS, Kitamura K, Hashimoto S, Ito H, Suzuki N, Sakai S, Morita M (2003). Polybrominated dibenzo-p-dioxins, dibenzofurans, and diphenyl ethers in Japanese human adipose tissue. *Environ Sci Technol*;37(5):817–21.

Chudler EH (2003). *Brain Facts* and *Figures*. Accessed Sept 2003 at: http://faculty.washington.edu/chudler/facts.html#brain

Corbellis L, Latini G, Felice CD, Razzi S, Paris I, Ruggieri F, Mazzeo P, Petraglia F (2003). High plasma concentrations of di-(2-ethylhexyl)-phthalate in women with endometriosis. *Hum Reprod*;18(7):1512–5.

Colon I, Caro D, Bourdony CJ, Rosario O (2000). Identification of phthalate esters in the serum of young Puerto Rican girls with premature breast development. *Environ Health Perspect*; 108 (9):895–900.

Dachs J, Van Ry DA, Eisenreich SJ (1999). Occurrence of estrogenic nonylphenols in the urban and coastal atmosphere of the lower Hudson river estuary. *Environ Sci Tech*;33:2676–9.

Darnerud PO, Eriksen GS, Johannesson T, Larsen PB, Viluksela M (2001). Polybrominated diphenyl ethers: Occurrence, dietary exposure, and toxicology. *Environ Health Perspect*;109(Suppl 1):49–68. Darnerud PO (2003). Toxic effects of brominated flame retardants in man and in wildlife. *Environ Int*;29(6):841–53.

Davis DL, Gottlieb MB, Stampnitzky JR (1998). Reduced ratio of male to female births in several industrial countries: A sentinel health indicator? *JAMA*;279(13):1018–23.

Doering DD, Steckelbroeck S, Doering T, Klingmuller D (2002). Effects of butyltins on human 5alpha-reductase type 1 and type 2 activity. *Steroids*;67(10):859–87.

Dostal LA, Weaver RP, Schwetz BA (1987). Transfer of di(2-ethylhexyl) phthalate through rat milk and effects on milk composition and the mammary gland. *Toxicol Appl Pharmacol*;**91**(3):315–25.

EC (2002). Opinion on Medical Devices Containing DEHP Plasticised PVC; Neonates and Other Groups Possibly at Risk from DEHP Toxicity. European Commission, Health & Consumer Protection Directorate-General. Accessed Sept 2003 at: http://europa.eu.int/comm/food/fs/sc/scmp/out43\_en.pdf

EEA (1999). Chemicals in the European Environment: Low Doses, High Stakes? Copenhagen, Denmark: European Environment Agency (EEA) and United Nations Environment Programme (UNEP). Accessed Sept 2003: http://reports.eea.eu.int/NYM2/en

EHP (1996). Statement from the work session on chemically-induced alterations in the develop-



ing immune system: The wildlife/human connection. *Environ Health Perspect*; 104 (Suppl 4):807–8.

Eichenfield LF, Hardaway CA (1999). Neonatal dermatology. *Curr Opin Pediat*;11:471–4.

Eisenhardt S, Runnebaum B, Bauer K, Gerhard I (2001). Nitromusk compounds in women with gynecological and endocrine dysfunction. *Environ Res*; 87 (3):123–30

ENDS (1999a). Industry glimpses new challenges as endocrine science advances. *The ENDS Report*;(290):26–30.

ENDS (1999b). Plastics contaminate tap water with hormone disrupters. *The ENDS Report*;(293):4–5.

Eriksson P, Nordberg A (1986). The effects of DDT, DDOH-palmitic acid, and a chlorinated paraffin on muscarinic receptors and the sodium-dependent choline uptake in the central nervous system of immature mice. *Toxicol Appl Pharmacol*;85(2):121–7.

Eriksson P, Jakobsson E, Fredriksson A (2001). Brominated flame retardants: A novel class of developmental neurotoxicants in our environment? *Environ Health Perspect*;109(9):903–8.

Foster W, Chan S, Platt L, Hughes C (2000). Detection of endocrine disrupting chemicals in samples of second trimester human amniotic fluid. *J Clin Endocrinol Metab*;85(8):2954–7.

Ginsberg G, Hattis D, Sonawane B, Russ A, Banati P, Kozlak M, Smolenski S, Goble R (2002). Evaluation of child/adult pharmacokinetic differences from a database derived from the therapeutic drug literature. *Toxicol Sci*;66(2):185–200.

Goldman LR (2002). Preventing pollution? U.S. toxic chemicals and pesticides policies and sustainable development. *Environmental Law Report News & Analysis*; 32. Washington, DC, USA: Environmental Law Institute. Accessed Sept 2003 at: http://www.cheforhealth.org/resources/lynngoldmanarticle.pdf

Gray LE Jr, Ostby J, Furr J, Price M, Veeramachaneni DN, Parks L (2000). Perinatal exposure to the phthalates DEHP, BBP, and DINP, but not DEP, DMP, or DOTP, alters sexual differentiation of the male rat. *Toxicol Sci*; 58(2):350–65.

Grech V, Vassallo-Agius P, Savona-Ventura C (2003). Secular trends in sex ratios at birth in North America and Europe over the second half of the 20th century. *J Epidemiol Community Health*;57(8):612–5.

Guenther, K, Heinke V, Thiele B, Kleist E, Prast H, Raecker T (2002). Endocrine disrupting nonylphenols are ubiquitous in food. *Environ Sci and Tech*;36(8):1676–80.

Gupta C (2000). Reproductive malformation of the male offspring following maternal exposure to estrogenic chemicals. *Proc Soc Exp Biol* 

Med;224(2):61-8.

Guvenius DM, Aronsson A, Ekman-Ordeberg G, Bergman A, Noren K (2003). Human prenatal and postnatal exposure to polybrominated diphenyl ethers, polychlorinated biphenyls, polychlorobiphenylols, and pentachlorophenol. *Environ Health Perspect*;111(9):1235–41.

Hallgren S, Darnerud PO (2002). Polybrominated diphenyl ethers (PBDEs), polychlorinated biphenyls (PCBs) and chlorinated paraffins (CPs) in rats-testing interactions and mechanisms for thyroid hormone effects. *Toxicology*;177(2-3):227–43.

Hardell L, Lindstrom G, van Bavel B, Wingfors H, Sundelin E, Liljegren G (1998). Concentrations of the flame retardant 2,2',4,4'-tetrabrominated diphenyl ether in human adipose tissue in Swedish persons and the risk for non-Hodgkin's lymphoma. *Oncol Res*;10(8):429–32.

Hattis D, Ginsberg G, Sonawane B, Smolenski S, Russ A, Kozlak M, Goble R (2003). Differences in pharmacokinetics between children and adults – II. Children's variability in drug elimination half-lives and in some parameters needed for physiologically-based pharmacokinetic modeling. *Risk Anal*;23(1):117–42.

Helleday T, Tuominen KL, Bergman A, Jenssen D (1999). Brominated flame retardants induce intragenic recombination in mammalian cells. *Mutat Res*;439(2):137–47.



Honma S, Suzuki A, Buchanan DL, Katsu Y, Watanabe H, Iguchi T (2002). Low dose effect of *in utero* exposure to bisphenol A and diethylstilbestrol on female mouse reproduction. *Reprod Toxicol*;16(2):117–22.

Hooper K, She J (2003). Lessons from the polybrominated diphenyl ethers (PBDEs): Precautionary principle, primary prevention, and the value of community-based body-burden monitoring using breast milk. *Environ Health Perspect*;111(1):109–14.

Hoppin JA, Brock JW, Davis BJ, Baird DD (2002). Reproducibility of urinary phthalate metabolites in first morning urine samples. *Environ Health Perspect*; 110 (5):515–8.

Houlihan J, Wiles R, Thayer K, Gray S (2003). Body burden. *The pollution in people*. Washington, DC, USA: Environmental Working Group. Accessed Sept 2003 at: http://www.ewg.org/reports/body-burden/

Hoyert DL, Arias E, Smith BL, Murphy SL, Kochaneck KD (2001). Deaths: Final data for 1999. *National Vital Statistics Report*;49(8);1–114. Hyattsville, MA, USA: National Centre for Health Statistics. Accessed Sept 2003 at:http://www.cdc.gov/nchs/data/nvsr/nvsr49/nvsr49\_08.pdf

Howdeshell KL, Hotchkiss AK, Thayer KA, Vandenbergh JG, vom Saal FS (1999). Exposure to bisphenol A advances puberty. *Nature*;401(6755):763–4. Huyghe E, Matsuda T, Thonneau P (2003). Increasing incidence of testicular cancer worldwide: A review. *J Urol*;170(1):5–11.

Ikeda H, Matsuyama S, Tanimura T (1997). Association between hepatoblastoma and very low birth weight: A trend or a chance? *J Pediatr*;130:557–60.

Ikezuki Y, Tsutsumi O, Takai Y, Kamei Y, Taketani Y (2002). Determination of bisphenol A concentrations in human biological fluids reveals significant early prenatal exposure. *Human Reprod*;17(11):2839–41.

Ikonomou MG, Rayne S, Addison RF (2002). Exponential increases of the brominated flame retardants, polybrominated diphenyl ethers, in the Canadian Arctic from 1981 to 2000. *Environ Sci Technol*;36(9):1886–92.

Inadera H, Sekiya T, Yoshimura T, Matsushima K (2000). Molecular analysis of the inhibition of monocyte chemoattractant protein-1 gene expression by estrogens and xenoestrogens in MCF-7 cells. *Endocrinology*;141(1):50–9.

Inoue K, Kato K, Yoshimura Y, Makino T, Nakazawa H (2000). Determination of bisphenol A in human serum by high-performance liquid chromatography with multi-electrode electrochemical detection. *J Chromatogr B Biomed Sci Appl*;749(1):17–23.

IPCS (1999). Tributyltin oxide. International Programme on Chemical Safety Concise International Chemical
Assessment Document: 14.
Accessed Sept 2003 at:
http://www.inchem.org/documents/cicads/cicads/cicad14.htm#PartNumber:14

Jacobson DL, Gange SJ, Rose NR, Graham NM (1997). Epidemiology and estimated population burden of selected autoimmune diseases in the United States. *Clin Immunol Immunopathol*; 84(3):223–43.

Jensen KG, Wiberg K, Klasson-Wehler E, Onfelt A (2000). Induction of aberrant mitosis with PCBs: Particular efficiency of 2, 3,3',4,4'-pentachlorobiphenyl and synergism with triphenyltin. *Mutagenesis*;15(1):9–15.

Kafferlein HU, Angerer J (2001). Trends in the musk xylene concentrations in plasma samples from the general population from 1992/1993 to 1998 and the relevance of dermal uptake. *Int Arch Occup Environ Health*; 74(7):470–6.

Kato Y, Kenne K (1996). Inhibition of cell-cell communication by commercial chlorinated paraffins in rat liver epithelial IAR 20 cells. *Pharmacol Toxicol*; 79(1):23–8.

Kawai K, Nozaki T, Nishikata H, Aou S, Takii M, Kubo C (2003). Aggressive behavior and serum testosterone concentration during the maturation process of male mice: The effects of fetal exposure to bisphenol A. *Environ Health Perspect*;111(2):175–8.



Kim JY, Jeong HG (2003). Down-regulation of inducible nitric oxide synthase and tumor necrosis factor-alpha expression by bisphenol A via nuclear factor-kappaB inactivation in macrophages. *Cancer Lett*;196(1):69–76.

Kleinsasser NH, Kastenbauer ER, Wallner BC, Weissacher H, Harreus UA (2001). Genotoxicity of phthalates. On the discussion of plasticizers in children's toys. [German]. *HNO*;49(5):378–81.

Koch HM, Drexler H, Angerer J (2003). An estimation of the daily intake of di(2-ethylhexyl)phthalate (DEHP) and other phthalates in the general population. *Int J Hyg Environ Health*;206(2):77–83.

Krstevska-Konstantinova M, Charlier C, Craen M, Du Caju M, Heinrichs C, de Beaufort C, Plomteux G, Bourguignon JP (2001). Sexual precocity after immigration from developing countries to Belgium: Evidence of previous exposure to organochlorine pesticides. *Hum Reprod*;16(5):1020–6.

Kuroda N, Kinoshita Y, Sun Y, Wada M, Kishikawa N, Nakashima K, Makino T, Nakazawa H (2003). Measurement of bisphenol A levels in human blood serum and ascitic fluid by HPLC using a fluorescent labeling

reagent. *J Pharm Biomed Anal*;**30**(6):1743–9.

Kyselova V, Peknicova J, Buckiova D, Boubelik M (2003). Effects of pnonylphenol and resveratrol on body and organ weight and *in vivo* fertility of outbred CD-1 mice. *Reprod Biol Endocrinol*; 1(1):30.

Larsen ST, Hansen JS, Thygesen P, Begtrup M, Poulsen OM, Nielsen GD (2001). Adjuvant and immuno-suppressive effect of six monophthalates in a subcutaneous injection model with BALB/c mice. Toxicology;169(1):37–51.

Lee HJ, Chattopadhyay S, Gong EY, Ahn RS, Lee K (2003a). Antiandrogenic effects of bisphenol A and nonylphenol on the function of androgen receptor. *Toxicol Sci*;75(1):40–6.

Lee MH, Chung SW, Kang BY, Park J, Lee CH, Hwang SY, Kim TS (2003b). Enhanced interleukin-4 production in CD4+ T cells and elevated immunoglobulin E levels in antigen-primed mice by bisphenol A and nonylphenol, endocrine disruptors: Involvement of nuclear factor-AT and Ca2+. *Immunology*;109(1):76–86.

Leisewitz A, Schwarz W (1998). Materials flow analysis of major endocrine disrupting industrial chemi-

cals (bisphenol A; dibutyl phthalate/benzyl butyl phthalate; nonylphenol/alkylphenol ethoxylates). Report on the Umweltbundesamt (German Federal Environmental Agency), UFOPLAN-No. 106 01 076. Accessed Sept 2003 at: http://www.oekorecherche.de / english/repseng.html

Liebl B, Ehrenstorfer S (1993). Nitro-musk compounds in breast milk. [German]. Gesundheitswesen;55(10):52 7–32.

Lo S, Allera A, Albers P, Heimbrecht J, Jantzen E, Klingmuller D, Steckelbroeck S (2003). Dithioerythritol (DTE) prevents inhibitory effects of triphenyltin (TPT) on the key enzymes of the human sex steroid hormone metabolism. *J Steroid Biochem Mol Biol*;84(5):569–76.

Loo TW, Clarke DM (1998). Nonylphenol ethoxylates, but not nonylphenol, are substrates of the human multidrug resistance P-glycoprotein. *Biochem Biophys Res*Commun;247(2):478–80.

Lopez-Cervantes J, Paseiro-Losada P (2003).
Determination of bisphenol A in, and its migration from, PVC stretch film used for food packaging. *Food Addit Contam*; 20(6):596–606.



Lovekamp TN, Davis BJ (2001). Mono-(2-ethylhexyl) phthalate suppresses aromatase transcript levels and estradiol production in cultured rat granulosa cells. *Toxicol Appl Pharmacol*; 172(3):217–24.

Lovekamp-Swan T, Davis BJ (2003). Mechanisms of phthalate ester toxicity in the female reproductive system. *Environ Health Perspect*;111(2):139–45.

Luster MI, Dean JH, Germolec DR (2003). Consensus workshop on methods to evaluate developmental immunotoxicity. Environ Health Perspect;111(4):579–83.

Mariussen E, Fonnum F (2003). The effect of brominated flame retardants on neurotransmitter uptake into rat brain synaptosomes and vesicles. *Neurochem Int*; 43(4–5):533–42.

Markey CM, Luque EH, Munoz de Toro M, Sonnenschein C, Soto AM (2001). *In utero* exposure to bisphenol A alters the development and tissue organization of the mouse mammary gland. *Biol Reprod*; 65:1215–23.

Maruyama K, Ikeda H, Koizumi T, Tsuchida Y (1999). Prenatal and postnatal histories of very low birthweight infants who developed hepatoblastoma. Pediatr Int;41(1):82-9.

Mazdai A, Dodder NG, Abernathy MP, Hites RA, Bigsby RM (2003). Polybrominated diphenyl ethers in maternal and fetal blood samples. *Environ Health Perspect*;111 (9):1249–52.

McCarthy J (2003). Drug metabolism and disposition in pediatric and gerontological stages of life. In: Craig CR, Stitzel RE (editors). *Modern Pharmacology with Clinical Applications* (6th Ed.). Philadelphia, PA, USA: Lippincott, Williams and Wilkins, p56–62.

Meek ME, Chan PKL (1994). Bis(2-ethylhexyl)phthalate: Evaluation of risks to health from environmental exposure in Canada. *Environ Carcin Ecotoxicol Rev*;C12:179–94.

Meerts IA, van Zanden JJ, Luijks EA, van Leeuwen-Bol I, Marsh G, Jakobsson E, Bergman A, Brouwer A (2000). Potent competitive interactions of some brominated flame retardants and related compounds with human transthyretin *in vitro*. *Toxicol Sci*;56(1):95–104.

Meironyte D, Noren K, Bergman A (1999). Analysis of polybrominated diphenyl ethers in Swedish human milk. A time-related trend study, 1972–1997. *J Toxicol Environ Health A*; 58(6):329-41.

Mersch-Sundermann V, Schneider H, Freywald C, Jenter C, Parzefall W, Knasmuller S (2001). Musk ketone enhances benzo(a)pyrene induced mutagenicity in human derived Hep G2 cells. *Mutat Res*;495(1–2):89–96.

Miura Y, Matsui H (1991). Inhibitory effects of phenyltin compounds on stimulus-induced changes in cytosolic free calcium and plasma membrane potential of human neutrophils. *Arch Toxicol*;65(7):562–9.

Moore KL, Persaud TVN (2003). The Developing Human. Clinically Orientated Embryology (7th Ed). Philadelphia, PA, USA: Saunders.

Moore RW, Rudy TA, Lin TM, Ko K, Peterson RE (2001). Abnormalities of sexual development in male rats with *in utero* and lactational exposure to the antiandrogenic plasticizer Di(2-ethylhexyl) phthalate. *Environ Health Perspect*; 109(3):229–37.

Nagel SC, vom Saal FS, Thayer KA, Dhar MG, Boechler M, Welshons WV (1997). Relative binding affinity-serum modified access (RBA-SMA) assay predicts the relative *in vivo* bioactivity of the xenoestrogens bisphenol A and



octylphenol. *Environ Health Perspect*; **105**(1):70–6

Nagel SC, vom Saal FS, Welshons WV (1999). Developmental effects of estrogenic chemicals are predicted by an *in vitro* assay incorporating modification of cell uptake by serum. *J Steroid Biochem Mol Biol*:69(1–6):343–57.

Nakata H, Sakakibara A, Kanoh M, Kudo S, Watanabe H, Nagai N, Miyazaki N, Asano Y, Tanabe S (2002). Evaluation of mitogen-induced responses in marine mammal and human lymphocytes by *invitro* exposure of butyltins and non-ortho coplanar PCBs. *Environ Pollut*;120(2):245–53.

Nencioni A, Wesselborg S, Brossart P (2003). Role of peroxisome proliferator-activated receptor gamma and its ligands in the control of immune responses. *Crit Rev Immunol*;23(1–2):1–13.

Norris WR (1994). Triphenyltin Acetate. International Programme on Chemical Safety Poisons Information Monograph: 589. Accessed Sept 2003 at: http://www.inchem.org/documents/pims/chemical/pim589.

NTP (1986a). NTP Toxicology and Carcinogenesis Studies of Chlorinated Paraffins (C23, 43% Chlorine) (CAS No. 108171-27-3) in F344/N Rats and B6C3F1 Mice (Gavage Studies). *Natl Toxicol Program Tech Rep Ser*;305:1–202.

NTP (1989b). NTP Toxicology and Carcinogenesis Studies of Chlorinated Paraffins (C12, 60% Chlorine) (CAS No. 108171-26-2\*) in F344/N Rats and B6C3F1 Mice (Gavage Studies). Natl Toxicol Program Tech Rep Ser:308:1–206.

Ohta S, Ishizuka D, Nishimura H, Nakao T, Aozasa O, Shimidzu Y, Ochiai F, Kida T, Nishi M, Miyata H (2002). Comparison of polybrominated diphenyl ethers in fish, vegetables, and meats and levels in human milk of nursing women in Japan. *Chemosphere*;46(5):68 9–96.

Ozaki A, Baba T (2003). Alkylphenol and bisphenol A levels in rubber products. Food Addit Contam; 20(1):92–8.

Okubo T, Suzuki T, Yokoyama Y, Kano K, Kano I (2003). Estimation of estrogenic and anti-estrogenic activities of some phthalate diesters and monoesters by MCF-7 cell proliferation assay *in vitro*. *Biol Pharm Bull*;26(8):1219–24.

Olsen CM, Meussen-Elholm ET, Samuelsen M, Holme JA, Hongslo JK (2003). Effects of the environmental oestrogens bisphenol A, tetrachlorobisphenol A, tetrabromobisphenol A, 4-hydroxy-biphenyl and 4,4'-dihydroxy-biphenyl on oestrogen receptor binding, cell proliferation and regulation of oestrogen sensitive proteins in the human breast cancer cell line MCF-7. *Pharmacol Toxicol*;92(4):180–8.

Palanza PL, Howdeshell KL, Parmigiani S, vom Saal FS (2002). Exposure to a low dose of bisphenol A during fetal life or in adulthood alters maternal behavior in mice. *Environ Health Perspect*;110(Suppl 3):415–22.

Paris F, Balaguer P, Terouanne B, Servant N, Lacoste C, Cravedi JP, Nicolas JC, Sultan C (2002). Phenylphenols, biphenols, bisphenol-A and 4-tertoctylphenol exhibit alpha and beta estrogen activities and antiandrogen activity in reporter cell lines. *Mol Cell Endocrinol*;193(1–2):43–9.

Parmar D, Srivastava SP, Srivastava SP, Srivastava SP, Seth PK (1985). Hepatic mixed function oxidases and cytochrome P-450 contents in rat pups exposed to di-(2-ethylhexyl)phthalate through mother's milk. *Drug Metab Dispos*;13(3):368–70.



Paulozzi LJ, Erickson D, Jackson RJ (1997). Hypospadias trends in two US surveillance systems. *Pediatrics*;100(5):831–4.

Paulozzi LJ (1999). International trends in rates of hypospadias and cryptorchidism. *Environ Health Perspect*; 107(4):297–302.

Poon R, Lecavalier P, Chan P, Viau C, Hakansson H, Chu I, Valli VE (1995). Subchronic toxicity of a medium-chain chlorinated paraffin in the rat. *J Appl Toxicol*;15(6):455–63.

Rahman F, Langford KH, Scrimshaw MD, Lester JN (2001). Polybrominated diphenyl ether (PBDE) flame retardants. *Sci Total Environ*;275(1–3):1–17.

Ramos JG, Varayoud J, Sonnenschein C, Soto AM, Munoz De Toro M, Luque EH (2001). Prenatal exposure to low doses of bisphenol A alters the periductal stroma and glandular cell function in the rat ventral prostate. *Biol Reprod*;65(4):1271–7.

Ramos JG, Varayoud J, Kass L, Rodriguez H, Costabel L, Munoz-De-Toro M, Luque EH (2003). Bisphenol A induces both transient and permanent histofunctional alterations of the hypothalamic-pituitary-gonadal axis in prenatally exposed male rats. *Endocrinology*;

144(7):3206-15.

Rice D, Barone S Jr (2000). Critical periods of vulnerability for the developing nervous system: Evidence from humans and animal models. *Environ Health Perspect*;108(Suppl3):511–33

Riedel J, Birner G, van Dorp C, Neumann HG, Dekant W (1999). Haemoglobin binding of a musk xylene metabolite in man. *Xenobiotica*;29(6):573–82.

Rimkus GG, Wolf M (1996). Polycyclic musk fragrances in human adipose tissue and human milk. *Chemosphere*; 33 (10):2033–43.

Sakaue M, Ohsako S, Ishimura R, Kurosawa S, Kurohmaru M, Hayashi Y, Aoki Y, Yonemoto J, Tohyama C (2001). Bisphenol-A affects spermatogenesis in the adult rat even at a low dose. *J Occup Health*;43:185–90.

Santillo D, Labunska I, Davidson H, Johnston P, Strutt M, Knowles O (2003). Consuming chemicals. Hazardous chemicals in house dust as an indicator of chemical exposure in the home. London, UK: Greenpeace Environmental Trust.

Schmeiser HH, Gminski R, Mersch-Sundermann V (2001). Evaluation of health risks caused by musk ketone. *Int J Hyg Environ Health*;**203**(4):293–9.

Schmid PP, Muller MD (1985). Trace level detection of chlorinated paraffins in biological and environmental samples, using gas chromatography/mass spectrometry with negative-ion chemical ionization. *J Assoc Off Anal Chem*; 68 (3):427–30.

Schonfelder G, Flick B, Mayr E, Talsness C, Paul M, Chahoud I (2002a). *In utero* exposure to low doses of bisphenol A lead to long-term deleterious effects in the vagina.

Neoplasia;4(2):98–102.

Schonfelder G, Wittfoht W, Hopp H, Talsness CE, Paul M, Chahoud I (2002b). Parent Bisphenol A accumulation in the human maternal–fetal–placental unit. *Environ Health Perspect*;110(11):A703–7.

Schreurs RH, Quaedackers ME, Seinen W, van der Burg (2002). Transcriptional activation of estrogen receptor ERalpha and ERbeta by polycyclic musks is cell type dependent. *Toxicol Appl Pharmacol*;183(1):1–9.

Schroter-Kermani (2001). Endocrine disrupters in human and environmental samples from the German Environmental Specimen Bank. *Proceedings of the* 



Second Status Seminar: Endocrine Disrupters. 2–4 April, 2001, Berlin, Germany.

Sharpe CR, Franco EL (1995). Etiology of Wilm's tumor. *Epidemiology Review*;17:415–32.

Sharpe RM, Fisher JS, Millar MM, Jobling S, Sumpter JP (1995). Gestational and lactational exposure of rats to xenoestrogens results in reduced testicular size and sperm production. *Environ Health Perspect*;103 (12):1136–43.

She J, Petreas M, Winkler J, Visita P, McKinney M, Kopec D (2002). PBDEs in the San Francisco Bay area: Measurements in harbor seal blubber and human breast adipose tissue. *Chemosphere*;46(5):697–707.

Shetler T, Stein J, Reich F, Walings D (2000). In Harm's Way: Toxic Threats to Child Development. Cambridge, MA, USA: Greater Boston Physicians for Social Responsibility. Accessed Sept 2003 at: http://psr.igc. org/ihw.htm#ProjPhases

Siderowf A, Stern M (2003). Update on Parkinson disease. Ann Intern Med; 138(8):651–8.

Skakkebaek NE, Rajpert-De Meyts, Main KM (2001). Testicular dysgenesis syndrome: An increasingly common developmental disorder with environmental aspects. *Hum Reprod*;16(5):972–8.

Sonnenschein C, Soto AM (1999). The Society of Cells: Cancer and Control of Cell Proliferation. New York, NY, USA: Springer Verlag.

Srivastava S, Awasthi VK, Srivastava SP, Seth PK (1989). Biochemical alterations in rat fetal liver following *in utero* exposure to di(2-ethylhexyl)phthalate (DEHP). *Indian J Exp Biol*;27(10):885–8.

Steingraber S (2001). Having Faith. An Ecologist's Journey to Motherhood. Oxford, UK: The Perseus Press.

Steckelbroeck S, Heidrich D, Heimbrecht J, Klingmuller D (2001). Effects of triphenyltin (TPT) on the key enzymes of the human sex steroid hormone metabolism. Proceedings of the Second Status Seminar: Endocrine Disrupters, 2–4 April 2001, Berlin Germany. Abstracts; p127.

Stridh H, Cotgreave I, Muller M, Orrenius S, Gigliotti D (2001). Organotin-induced caspase activation and apoptosis in human peripheral blood lymphocytes. *Chem Res Toxicol*;14(7):791–8.

Swan SH, Elkin EP, Fenster L (2000). The question of

declining sperm density revisited: An analysis of 101 studies published 1934–1996. Environ Health Perspect; 108(10):961–6.

Takada H, Isobe T, Nakada N, Nishiyama H, Iguchi T, Irie H, Mori C (1999). Bisphenol A and nonylphenols in human umbilical cords. Proceedings of the International Scientific Conference on Environmental Endocrine Disrupting Chemicals, Monte Verita, Ascona, Switzerland, March 7–12, 1999.

Takahashi S, Mukai H, Tanabe S, Sakayama K, Miyazaki T, Masuno H (1999). Butyltin residues in livers of humans and wild terrestrial mammals and in plastic products. *Environ Pollution*;106(2):213–8.

Takai Y, Tsutsumi O, Ikezuki Y, Hiroi H, Osuga Y, Momoeda M, Yano T, Taketani Y (2000). Estrogen receptor-mediated effects of a xenoestrogen, bisphenol A, on preimplantation mouse embryos. *Biochem Biophys ResCommun*;270(3):918–21.

Takai Y, Tsutsumi O, Ikezuki Y, Kamei Y, Osuga Y, Yano T, Taketan Y (2001). Preimplantation exposure to bisphenol A advances postnatal development. *Reprod Toxicol*;15(1):71–4.

Takeuchi T, Tsutsumi O (2002). Serum bisphenol A



concentrations showed gender differences, possibly linked to androgen levels. *Biochem Biophys Res Commun*;**291**(1):76–8.

Thomsen C, Lundanes E, Becher G (2002). Brominated flame retardants in archived serum samples from Norway: A study on temporal trends and the role of age. *Environ Sci Technol*;36(7):1414–8.

Tsumura Y, Ishimitsu S, Saito I, Sakai H, Kobayashi Y, Tonogai Y (2001a). Eleven phthalate esters and di(2-ethylhexyl) adipate in one-week duplicate diet samples obtained from hospitals and their estimated daily intake. *Food Addit Contam*; 18(5):449–60.

Tsumura Y, Ishimitsu S, Kaihara A, Yoshii K, Nakamura Y, Tonogai Y (2001b). Di(2-ethylhexyl) phthalate contamination of retail packed lunches caused by PVC gloves used in the preparation of foods. *Food Addit Contam*;18(6):569–79.

Uchida K, Suzuki A, Kobayashi Y, Buchanan DL, Sato T, Watanabe H, Katsu Y, Suzuki J, Asaoka K, Mori C, Arizono K, Iguchi T (2002). Bisphenol-A administration during pregnancy results in female exposure in mice and monkeys. *J Health Sci*;48:579–82. US EPA (1993). Bisphenol A (CASRN 80-05-7). Reference dose for chronic oral exposure (RfD). Last revised in 1993. *Integrated Risk Information System*. US Environment Protection Agency. Accessed Sept 2003 at: http://www.epa.gov/iris/subst/0356.htm

US EPA (2003a). 2001
Toxics Release Inventory
(TRI). Public Data Release.
Executive Summary.
Washington, DC, USA:
Environmental Protection
Agency. Paper number: EPA
260-S-03-001. Accessed Sept
2003 at: http://www.epa.gov/
tri/tridata/tri01/press/executivesummarystandalone.pdf

US EPA (2003b). America's Children and the Environment: Measures of Contaminants, Body Burdens, and Illnesses.
Washington, DC, USA:
United States Environmental Protection Agency. Paper number: EPA 240-R0-3001http://yosemite.epa.gov/ochp/ochpweb.nsf/content/pu blications.htm

van Beerendonk GJ, Nelson SD, Meerman JH (1994). Metabolism and genotoxicity of the halogenated alkyl compound tris(2,3-dibromopropyl)phosphate. *Hum Exp Toxicol*;13(12):861–5.

van Heijst APN (1994). Tributyltin compounds. International Programme on Chemical Safety Group Poisons Information Monograph: G018. Accessed Sept 2003 at: http://www.inchem.org/ documents/pims/chemical/pi mg018.htm

Viberg H, Fredriksson A, Jakobsson E, Orn U, Eriksson P (2003). Neurobehavioural derangements in adult mice receiving decabrominated diphenyl ether (PBDE 209) during a defined period of neonatal brain development. *Toxicol Sci*; Aug 12, 2003 [Epub ahead of print]. Accessed Sept 2003 at: http://www.toxsci.oupjournals.org/cgi/content/abstract/kfg210v1

vom Saal FS, Cooke PS, Buchanan DL, Palanza P, Thayer KA, Nagel SC, Parmigiani S, Welshons WV (1998). A physiologically based approach to the study of bisphenol A and other estrogenic chemicals on the size of reproductive organs, daily sperm production, and behavior. *Toxicol Ind Health*;14(1–2):239–60.

Warngard L, Bager Y, Kato Y, Kenne K, Ahlborg UG (1996). Mechanistical studies of the inhibition of intercellular communication by organochlorine compounds. *Arch Toxicol Suppl*; 18:149–59.

Watanabe H, Adachi R, Hirayama A, Kasahara T, Suzuki K (2003).



Triphenyltin enhances the neutrophilic differentiation of promyelocytic HL-60 cells. *Biochem Biophys Res Commun*;306(1):26–31.

Wetherill YB, Petre CE, Monk KR, Puga A, Knudsen KE (2002). The xenoestrogen bisphenol A induces inappropriate androgen receptor activation and mitogenesis in prostatic adenocarcinoma cells. *Mol Cancer Ther*;1(7):515–24.

Whalen MM, Loganathan BG (2001). Butyltin exposure causes a rapid decrease in cyclic AMP levels in human lymphocytes. *Toxicol Appl Pharmacol*;171(3):141–8.

Whalen MM, Green SA, Loganathan BG (2002). Brief butyltin exposure induces irreversible inhibition of the cytotoxic function on human natural killer cells, *in vitro*. *Environ Res*;88(1):19–29.

Whalen MM, Walker L, Loganathan BG (2002a). Interleukins 2 and 12 produce significant recovery of cytotoxic function in dibutyltin-exposed human natural killer cells. *Environ Res*;88(2):103–15.

Whalen MM, Williams TB, Green SA, Loganathan BG (2002b). Interleukins 2 and 12 produce recovery of cytotoxic function in tributyltin-exposed human natural killer cells. *Environ Res*;88(3):199–209.

Whalen MM, Wilson S, Gleghorn C, Loganathan BG (2003). Brief exposure to triphenyltin produces irreversible inhibition of the cytotoxic function of human natural killer cells. *Environ Res*;92(3):213–20.

WHO (2000). Bronchial Asthma. WHO Fact Sheet No. 206. Accessed Sept 2003 at: http://www.who.int/inf-fs/en/fact206.html

Wilson NK, Chuang JC, Lyu C, Menton R, Morgan MK (2003). Aggregate exposures of nine preschool children to persistent organic pollutants at day care and at home. *J Expo Anal Environ Epidemiol*;13(3):187–202.

Woodward G (2001). Autism and Parkinson's disease. *Med Hypotheses*;56(2):246–9.

Yamada H, Furuta I, Kato EH, Kataoka S, Usuki Y, Kobashi G, Sata F, Kishi R, Fujimoto S (2002). Maternal serum and amniotic fluid bisphenol A concentrations in the early second trimester. *Reprod Toxicol*;16(6):735–9.

Yamasaki H, Nagake Y, Makino H (2001). Determination of bisphenol A in effluents of hemodialyzers. *Nephron*;88(4):376–8. Yamashita U, Sugiura T, Yoshida Y, Kuroda E (2003). Effect of endocrine disrupters on thymocytes *in vitro*. *JUOEH*;25(2):161–70.

Ying GG, Williams B, Kookana R (2002). Environmental fate of alkylphenols and alkylphenol ethoxylates – A review. *Environ Int*;28(3):215–26.

Zehringer M, Herrmann A (2001). Analysis of polychlorinated biphenyls, pyrethroid insecticides and fragrances in human milk using a laminar cup liner in the GC injector. *Eur Food Res Tech*; 212:247–51.

October 2003

GREENPEACE

Canonbury Villas London N1 2PN UK

www.greenpeace.org.uk/toxics